NBER WORKING PAPER SERIES

GOODS TRADE, FACTOR MOBILITY AND WELFARE

Stephen J. Redding

Working Paper 18008 http://www.nber.org/papers/w18008

NATIONAL BUREAU OF ECONOMIC RESEARCH 1050 Massachusetts Avenue Cambridge, MA 02138 April 2012

I am grateful to Kevin Lim and Tomasz Swiecki for research assistance and Princeton University for research support. I would also like to thank seminar participants at the American Economic Association (AEA), Princeton University and co-authors and colleagues for helpful comments and suggestions, including Lorenzo Caliendo, Swati Dhingra, Dave Donaldson, Gene Grossman, Elhanan Helpman, Sam Kortum, Dan Lu, Marc Melitz, Guy Michaels, Ralph Ossa, Gianmarco Ottaviano, Ferdinand Rauch, Esteban Rossi-Hansberg and Daniel Sturm. The usual disclaimer applies. The views expressed herein are those of the author and do not necessarily reflect the views of the National Bureau of Economic Research.

NBER working papers are circulated for discussion and comment purposes. They have not been peerreviewed or been subject to the review by the NBER Board of Directors that accompanies official NBER publications.

© 2012 by Stephen J. Redding. All rights reserved. Short sections of text, not to exceed two paragraphs, may be quoted without explicit permission provided that full credit, including © notice, is given to the source.

Goods Trade, Factor Mobility and Welfare Stephen J. Redding NBER Working Paper No. 18008 April 2012. 'Tgxkugf 'Lwpg''4236 JEL No. F11,F12,F16

ABSTRACT

Much trade occurs intra-nationally across regions within countries, where not only goods but also'factors of production are mobile across locations. This paper extends a recent class of quantitative'trade models to incorporate factor mobility. We generalize the standard approach to undertaking counterfactuals in these models to allow for endogenous reallocations of mobile factors across locations. We show that each location's trade share with itself is no longer a sufficient statistic for the welfare gains from trade. We derive a bias correction term for adjusting the standard measure of the aggregate welfare gains from trade to take into account these endogenous reallocations of resources.

Stephen J. Redding Department of Economics and Woodrow Wilson School Princeton University Fisher Hall Princeton, NJ 08544 and NBER reddings@princeton.edu

1 Introduction

The theoretical result that there are gains from trade is a central tenet of economics. Assuming perfect competition and no market failures, trade acts like a technological improvement that expands the set of feasible allocations and enables Pareto superior outcomes to be achieved. A recent body of research has quantified the magnitude of these welfare gains in a class of general equilibrium models of trade. A common assumption in this line of research has been the immobility of factors of production across locations. While this assumption is arguably a reasonable approximation for international trade between countries, much trade occurs intra-nationally between regions within countries, where at least some factors of production are likely to be mobile across locations. Furthermore, this internal mobility of factors across regions within countries can itself influence the effects of external reductions in trade costs between countries.¹

We extend this class of recent quantitative trade models to incorporate factor mobility across regions within countries. We show that the standard conditions for general equilibrium in these models are augmented by an additional relationship that determines the distribution of the mobile factor across regions. Using the structure of the model and data on factor supplies and trade shares, we show that the dispersion of mobile factors can be decomposed into the contributions of locational fundamentals, market access and their covariance.

We generalize the standard approach to undertaking counterfactuals in quantitative trade models to allow for factor mobility. The key new feature of this generalized approach is that external trade liberalization leads to endogenous changes in the distribution of economic activity across regions within countries. We first demonstrate these results in settings in which both goods and factors are mobile across locations. We next generalize the analysis to a setting in which goods are mobile both across regions within countries and across countries, whereas factors of production are only mobile across regions within countries but not across countries.

We show that endogenous reallocations of the mobile factor across regions within countries have ramifications for the representative agent's welfare gains from trade. A location's trade share with itself is no longer a sufficient statistic for the welfare gains from trade, which also depend on endogenous changes in the supply of the mobile factor. We derive a bias correction term for the standard measure of a country's aggregate welfare gains from trade that adjusts for these internal reallocations of resources. In a quantitative application to trade between Canadian provinces and U.S. states, we show that endogenous changes in the distribution of mobile factors within countries are quantita-tively relevant for evaluating the representative agent's welfare gains from trade.

Our analysis proceeds as follows. We begin by developing our results in a simple neoclassical framework following Eaton and Kortum (2002), in which an economy of many regions is linked through both goods trade and factor mobility. Our results are, however, more general and hold within a wider class of models that has many features in common with the class of models consid-

¹For evidence on the importance of trade within countries, see for example Donaldson (2014) and Hillberry and Hummels (2008). For evidence on factor mobility, see Blanchard and Katz (1992) and Bound and Holzer (2000).

ered by Arkolakis et al. (2012). These features include Dixit-Stiglitz preferences, perfect or monopolistic competition, balanced trade, aggregate profits that are a constant share of aggregate revenues (possibly zero), and CES import demand. The key difference in our analysis is the presence of both mobile and immobile factors of production, which introduces reallocations of mobile factors as a new channel through which international trade in goods can affect welfare. This new channel influences the overall magnitude of the welfare gains from trade, which can be no longer inferred from the change in a location's trade share with itself. To illustrate the broader applicability of our results, we show that they also hold within the Helpman (1998) model of new economic geography, which introduces agglomeration forces through the combination of love of variety preferences, transport costs and increasing returns to scale.

To develop our argument as clearly as possible, we initially consider settings in which there is a single immobile factor (land) that enters consumption (residential land use), a single mobile factor of production (labor), and all regions are linked through both goods trade and factor mobility. We begin by assuming that the mobile factor is perfectly mobile before extending the analysis to allow for imperfect mobility. We next generalize our analysis to allow the immobile factor to enter production (commercial land use), to introduce intermediate inputs, and to consider an economy of multiple countries in which factors are mobile across regions within countries but immobile between countries. We use this augmented framework as the basis for an empirical analysis of trade between Canadian provinces and U.S. states. Using data from Anderson and Van Wincoop (2003), we show that allowing for factor mobility within countries leads to quite different estimates of the welfare effects of changes in barriers to goods trade.

Our paper is most closely related to recent quantitative models of international trade following Eaton and Kortum (2002), including Alvarez and Lucas (2007), Arkolakis et al. (2012), Caliendo and Parro (2012), Costinot et al. (2012), Donaldson (2014), Eaton et al. (2011a, 2011b), Fieler (2011), Hsieh and Ossa (2011), Ossa (2011) and Simonovska and Waugh (2013). In contrast to these studies, we introduce factor mobility across locations, and show how it influences the predictions of these models for the general equilibrium effects of goods trade on wages, prices and welfare. Ramondo et al. (2014) show that quantitatively explaining the relationship between income per capita and country size in an extended Eaton and Kortum (2002) model requires multiple regions within countries. In contrast to their assumption of symmetric regions within countries, we allow regions to differ generally in terms of their locational fundamentals (productivity and land supply) and trade costs (both within and between countries). We show that reallocations across these asymmetric regions within countries are consequential for the representative agent's welfare gains from trade.

Our research is also related to the economic geography literature following Krugman (1991a,b), including Helpman (1998), Hanson (2005) and Redding and Sturm (2008).² Allen and Arkolakis (2013) generalize the Armington model of trade to incorporate factor mobility and provide general

²See also Davis and Weinstein (2002), Desmet and Rossi-Hansberg (2014), Fujita et al. (1999), Hanson (1996, 1997), Head and Ries (2001), Redding and Venables (2004), and Rossi-Hansberg (2005).

conditions for the existence, uniqueness and stability of equilibrium. In this Armington setting, the differentiation of goods by location of origin provides a dispersion force. In contrast, in our analysis, the immobility of land provides the dispersion force. This difference proves to be consequential for the measurement of the welfare gains from trade, which depend on endogenous reallocations of the mobile factor in our setting (through the demand for land), whereas they do not in an Armington framework without the addition of external economies. More generally, we relax the assumption that factors of production are perfectly mobile between all regions to allow for imperfect factor immobility and to allow factors of production to be mobile across regions within countries but not across countries. Coşar and Fajgelbaum (2013) develop a model of trade and internal geography, in which comparative advantage is similar across regions within countries, and regions completely specialize across tradable sectors. In contrast, we allow regions to differ from one another in terms of their comparative advantages, land supplies and trade costs, which gives rise to trade between regions within countries as well as between countries.

Finally, we build on a longer tradition in international trade that has examined the extent to which goods and factor movements are complements or substitutes, as in Markusen (1983), Mundell (1957) and Jones (1967), as well as work that has considered the role of the geography within countries in influencing international trade, such as Courant and Deardorff (1992, 1993) and Rauch (1991). We introduce factor mobility into a recent class of quantitative models of international trade and examine its implications for model-based counterfactuals and the welfare gains from trade.

The remainder of the paper is structured as follows. Section 2 introduces the baseline model that illustrates the role of factor mobility. Section 3 shows that these results hold in a wider class of models, including those featuring endogenous agglomeration forces. Section 4 generalizes the analysis to allow the mobile factor to be imperfectly mobile. Section 5 considers a more general setting with factor mobility within but not across countries. Section 6 illustrates the quantitative relevance of factor mobility for the welfare effects of changes in trade costs using data on U.S. states and Canadian Provinces. Section 7 concludes.

2 Theoretical Framework

We consider an economy consisting of a set N of regions indexed by n. Each region is endowed with an exogenous quality-adjusted supply of land (H_i). The economy as a whole is endowed with a measure \overline{L} of workers, where each worker has one unit of labor that is supplied inelastically with zero disutility. Workers are perfectly mobile across regions and hence in equilibrium real wages are equalized across all populated regions. Regions are connected by a bilateral transport network that can be used to ship goods such that $d_{ni} \ge 1$ units must be shipped from region i in order for one unit to arrive in region n.³

³Since our analysis uses patterns of bilateral trade to reveal information about unobserved bilateral trade frictions, we can allow for general variation in $d_{ni} \ge 1$ between pairs of locations *n* and *i* and are not required to impose $d_{ii} = 1$.

2.1 Consumer Preferences

Preferences are defined over goods consumption (C_n) and residential land use (H_{Un}) and are assumed to take the Cobb-Douglas form:⁴

$$U_n = \left(\frac{C_n}{\alpha}\right)^{\alpha} \left(\frac{H_{Un}}{1-\alpha}\right)^{1-\alpha}, \qquad 0 < \alpha < 1.$$
(1)

The goods consumption index is defined over consumption of a fixed continuum of goods $j \in [0, 1]$:

$$C_n = \left[\int_0^1 c_{nj}^{\rho} dj\right]^{1/\rho}, \qquad \sigma = \frac{1}{1-\rho}.$$
 (2)

2.2 Production

Each location draws an idiosyncratic productivity z(j) for each good j. Productivity is independently drawn across goods and locations from a Fréchet distribution:

$$F_i(z) = e^{-T_i z^{-\theta}},\tag{3}$$

where the scale parameter T_i determines average productivity for location *i* and the shape parameter θ controls the dispersion of productivity across goods.

Goods are homogeneous in the sense one unit of a given good is the same as any other unit of that good. Each good is produced with labor under conditions of perfect competition according to a linear technology.⁵ The cost to a consumer in location n of purchasing one unit of good j from location i is therefore:

$$p_{ni}(j) = \frac{d_{ni}w_i}{z_i(j)},\tag{4}$$

where w_i denotes the wage in location *i*.

2.3 Expenditure Shares and Price Indices

The representative consumer in location *n* sources each good from the lowest-cost supplier to that location:

$$p_n(j) = \min\{p_i(j); i \in N\}.$$
 (5)

Using equilibrium prices (4) and the properties of the Fréchet distribution following Eaton and Kortum (2002), the share of expenditure of region *n* on goods produced by region *i* is:

$$\pi_{ni} = \frac{T_i \left(d_{ni} w_i \right)^{-\theta}}{\sum_{s \in N} T_s \left(d_{ns} w_s \right)^{-\theta}},\tag{6}$$

⁴For empirical evidence using U.S. data in support of the constant housing expenditure share implied by the Cobb-Douglas functional form, see Davis and Ortalo-Magne (2011).

⁵While to simplify the exposition we begin by assuming that land is only used residentially, this assumption is straightforward to relax, as shown in Section 5 below.

while the price index dual to (2) can be expressed as:

$$P_n = \gamma \left[\sum_{i \in N} T_i \left(d_{ni} w_i \right)^{-\theta} \right]^{-1/\theta},$$
(7)

where $\gamma = \left[\Gamma\left(\frac{\theta+1-\sigma}{\theta}\right)\right]^{\frac{1}{1-\sigma}}$ and $\Gamma(\cdot)$ denotes the Gamma function.

Using the expenditure share (6), the price index (7) can be also written as:

$$P_n = \gamma \left(\frac{T_n}{\pi_{nn}}\right)^{-\frac{1}{\theta}} w_n,\tag{8}$$

so that locations with higher wages and higher trade shares with themselves have higher consumer goods price indices.

2.4 Income and Population Mobility

Expenditure on land in each location is redistributed lump sum to the workers residing in that location, as in Helpman (1998). Therefore total income in each location (v_n) equals labor income plus expenditure on residential land:

$$v_n L_n = w_n L_n + (1 - \alpha) v_n L_n = \frac{w_n L_n}{\alpha}.$$
(9)

Labor income in each location equals expenditure on goods produced in that location:

$$w_i L_i = \sum_{n \in \mathbb{N}} \pi_{ni} w_n L_n. \tag{10}$$

Land market clearing implies that the equilibrium land rent can be determined from the equality of land income and expenditure:

$$r_n = \frac{(1-\alpha)v_n L_n}{H_n} = \frac{1-\alpha}{\alpha} \frac{w_n L_n}{H_n}.$$
(11)

Population mobility implies that workers receive the same real income in all populated locations:

$$V_n = \frac{v_n}{P_n^{\alpha} r_n^{1-\alpha}} = \bar{V},\tag{12}$$

where r_n denotes the land rent in location n.

Using land market clearing (11), the equality of income and expenditure (9) and the price index (8), the above population mobility condition can be used to solve for the equilibrium population of each location as a function of locational fundamentals (productivity (T_n) and land quality (H_n)), the location's trade share with itself (π_{nn}) and the common level of real income (\bar{V}):

$$L_n = \frac{\left(\frac{T_n}{\pi_{nn}}\right)^{\overline{\theta(1-\alpha)}} H_n}{\alpha^{\frac{1-\alpha}{1-\alpha}} \left(\frac{1-\alpha}{\alpha}\right) \gamma^{\frac{\alpha}{1-\alpha}} \overline{V}^{\frac{1}{1-\alpha}}},$$
(13)

where labor market clearing requires:

$$\sum_{n \in N} L_n = \bar{L}.$$
 (14)

The expression for equilibrium population (13) has an intuitive interpretation. Locations with higher productivity (T_n) pay higher wages (w_n), which attracts population and bids up the price of land until real wages are equalized. Locations with higher land quality (H_n) have lower quality-adjusted prices of land, which again attracts population and bids up the price of land until real wages are equalized. Finally, locations that have low equilibrium trade shares with themselves (π_{nn}) for given values of productivity (T_n) and land quality (H_n) have low trade costs to other locations (high market access). These low trade costs to other locations imply low price indices for tradable goods (P_n), which must be offset in equilibrium by a higher population that bids up the price of land to achieve real wage equalization. Note that locations with high productivity (T_n) have high trade shares with themselves (π_{nn}), other things equal, both because higher productivity increases a location's trade share with itself in (6), and also because higher productivity increases a location's population, which reduces its wage relative to a weighted average of wages in other locations in (10).

2.5 General Equilibrium

The general equilibrium of the model can be represented by the share of workers in each location $(\lambda_n = L_n/\bar{L})$, the share of each location's expenditure on goods produced in other locations (π_{ni}) and the wage in each location (w_n) . Using labor income (10), the trade share (6), population mobility (13) and labor market clearing (14), this equilibrium triple $\{\lambda_n, \pi_{ni}, w_n\}$ solves the following system of equations for all $i, n \in N$:

$$w_i \lambda_i = \sum_{n \in N} \pi_{ni} w_n \lambda_n, \tag{15}$$

$$\pi_{ni} = \frac{T_i \left(d_{ni} w_i\right)^{-\theta}}{\sum_{k \in N} T_k \left(d_{nk} w_k\right)^{-\theta}},\tag{16}$$

$$\lambda_n = \frac{\left(\frac{T_n}{\pi_{nn}}\right)^{\frac{\alpha}{\theta(1-\alpha)}} H_n}{\sum_{k \in N} \left(\frac{T_k}{\pi_{kk}}\right)^{\frac{\alpha}{\theta(1-\alpha)}} H_k}.$$
(17)

Proposition 1 Given locational fundamentals (productivity (T_n) , land quality (H_n) and bilateral trade frictions (d_{ni})), there exist equilibrium population shares (λ_n) , trade shares (π_{ni}) and wages (w_n) that solve the wage equation (15), trade equation (16) and population mobility condition (17).

Proof. See the appendix.

The proposition establishes the existence of equilibrium as follows. First, for given population shares (λ_i), the wage equation (15) and trade equation (16) determine equilibrium wages (w_i) and trade shares (π_{ni}) for each location. Second, the existence of equilibrium population shares can be established by using the population mobility condition (17) together with the wage equation (15) and trade equation (16), as shown formally in the proof of the proposition.

In Figure 1, we illustrate the determination of equilibrium graphically for the special case of two regions {*A*, *B*}, which is also considered more formally in the appendix. In this case, $\lambda_B = 1 - \lambda_A$

and we choose the wage in location A as the numeraire. The figure shows the left and right-hand sides of the population mobility condition (17) on the *y*-axis against population share (λ_A) on the *x*axis. The left-hand side of (17) for region A is shown by the ray from the origin with slope 45 degrees, while the right-hand side corresponds to the downward-sloping curve. As location A's population share converges towards zero ($\lim_{\lambda_A \to 0}$), the wage equation (15) and trade equation (16) imply that its trade share with itself converges towards zero ($\pi_{AA} \to 0$), which in turn implies that the right-hand side of the population mobility condition (17) converges towards infinity. Conversely, as location A's population share converges towards one ($\lim_{\lambda_A \to 1}$), the wage equation (15) and trade equation (16) imply that the trade share of location B with itself converges towards zero ($\pi_{BB} \to 0$), which in turn implies that the right-hand side of the population mobility condition (17) for region A converges towards zero. Finally, the wage equation (15), trade equation (16) and population mobility condition (17) are continuous in λ_A , and hence there exist equilibrium population shares { $\hat{\lambda}_A$, $1 - \hat{\lambda}_A$ }. Having established the existence of these equilibrium population shares, equilibrium wages { $1, \psi_B$ } and trade shares { $\hat{\pi}_{AA}, \hat{\pi}_{AB}, \hat{\pi}_{BB}, \hat{\pi}_{BA}$ } can be determined from the wage equation (15) and trade equation (16).

Proposition 2 Given observed data on population (L_n) and trade shares (π_{ni}) , the population mobility condition (13) can be used to determine a composite measure of fundamentals for each location that incorporates productivity and land quality $(H_n T_n^{\alpha/\theta(1-\alpha)})$. Using this measure, the variance of population across locations can be decomposed into the contributions of fundamentals $(H_n T_n^{\alpha/\theta(1-\alpha)})$, market access (π_{nn}) and their covariance.

Proof. The proposition follows immediately from the population mobility condition (13) given observed data on population (L_n) and trade shares (π_{ni}):

$$\operatorname{var} \left\{ \log L_n \right\} = \operatorname{var} \left\{ \log \left(T_n^{\alpha/\theta(1-\alpha)} H_n \right) \right\} + \left(\frac{\alpha}{\theta(1-\alpha)} \right)^2 \operatorname{var} \left\{ \log \pi_{nn} \right\} \\ -2 \left(\frac{\alpha}{\theta(1-\alpha)} \right) \operatorname{covar} \left\{ \log \left(T_n^{\alpha/\theta(1-\alpha)} H_n \right), \log \left(\pi_{nn} \right) \right\}.$$

As summarized in the above proposition, the model yields a decomposition of the variance of population across locations into the contributions of locational fundamentals, market access and their covariance. Locational fundamentals are captured by a composite measure that incorporates productivity (T_n) and land quality (H_n). Market access is summarized by a location's trade share with itself (π_{nn}), where a lower own trade share implies lower relative trade costs to other locations (greater market access). One key implication of the population mobility condition (13) is that a composite measure of unobserved locational fundamentals ($H_n T_n^{\alpha/\theta(1-\alpha)}$) can be obtained from the model using only data on population (L_n) and trade shares (π_{nn}) and given a normalization of the common level of utility \bar{V} . To measure these locational fundamentals, we are not required to make assumptions about the form of bilateral trade costs (d_{ni}), which are instead implicitly revealed by observed bilateral trade flows, as reflected in each location's trade share with itself (π_{nn}).



Figure 1: Equilibrium Population Shares $\{\lambda_A, 1 - \lambda_A\}$ for the Case of Two Regions

2.6 Welfare Gains from Trade

Population mobility implies that the representative agent obtains the same welfare across all locations. Having solved for the equilibrium triple { λ_n , π_{ni} , w_n }, the price index (7) follows immediately from equilibrium wages. Using population mobility (13), the welfare of the representative agent in each location can be expressed as:

$$V_n = \frac{\left(\frac{T_n}{\pi_{nn}}\right)^{\frac{\alpha}{\theta}} H_n^{1-\alpha}}{\alpha \left(\frac{1-\alpha}{\alpha}\right)^{1-\alpha} \gamma^{\alpha} L_n^{1-\alpha}} = \bar{V},$$
(18)

where terms in wages (w_n) have canceled and land quality (H_n) is a parameter.

Proposition 3 *The change in a location's trade share with itself and the change in its population are sufficient statistics for the welfare gains from goods trade.*

Proof. The proposition follows immediately from (18) since:

$$\frac{V_n^{\text{Trade}}}{V_n^{\text{No Trade}}} = \left(\pi_{nn}^{\text{Trade}}\right)^{-\frac{\alpha}{\theta}} \left(\frac{L_n^{\text{Trade}}}{L_n^{\text{No Trade}}}\right)^{-(1-\alpha)} = \frac{\bar{V}^{\text{Trade}}}{\bar{V}^{\text{No Trade}}},$$

where $\pi_{nn}^{\text{No Trade}} = 1$.

In Proposition 3, we consider a reduction in barriers to trade in goods between locations that are linked through factor mobility. As a result of this factor mobility, the welfare gains from trade cannot be inferred simply from the change in each location's trade share with itself, but also depend on the change in each location's population. The reason is that a larger population raises the demand for land, which in turn increases the price of land and hence reduces welfare. Indeed, the model implies that the changes in population and trade shares induced by goods market integration are systematically related to one another. Suppose that we start from an equilibrium in which population

mobility has equalized real wages and that there is a reduction in barriers to trade in goods that is uneven across locations (e.g. coastal locations with lower geographic barriers to trade benefit disproportionately from the opening of closed economies to trade). Those locations that experience larger reductions in their trade share with themselves as a result of this goods market integration become more attractive locations, and hence experience larger increases in their population until real wages are again equalized in the new equilibrium following the reduction in trade barriers.

Our framework satisfies many of the conditions specified in Arkolakis et al. (2012) for a location's trade share with itself to be a sufficient statistic for the welfare gains from trade. In particular, our framework features Dixit-Stiglitz preferences, linear cost functions, perfect or monopolistic competition, balanced trade, aggregate profits are a constant share of aggregate revenues (zero), and CES import demand. The one condition violated is a single factor of production. Indeed, it is the combination of a mobile factor (labor) and immobile factor (land) that is central to our result in Proposition 3. To the extent that some factors of production are mobile across locations, our analysis highlights an interaction between trade in goods and factor allocations. This interaction will also prove important below when we consider the effects of goods market integration in a setting where factors are mobile across regions within countries but immobile across countries.

While the combination of mobile and immobile factors opens up a new channel through which goods market integration affects welfare, the presence of an immobile factor also mechanically dampens the overall magnitude of the welfare gains from trade in goods, because these welfare gains are only realized for tradable consumption goods. As a result, the exponent on a location's trade share with itself (π_{nn}) in welfare (18) is the share of tradable goods in consumption (α) divided by the elasticity of trade flows with respect to trade costs (θ).

2.7 Counterfactuals

As in the recent class of quantitative trade theories, the system of equations for general equilibrium (15)-(17) can be used to undertake model-based counterfactuals following the methodology introduced by Dekle, Eaton and Kortum (2005). In contrast to these recent quantitative trade theories, our framework yields predictions for endogenous changes in population across locations, which in turn feedback to influence wages, price indices and trade patterns between regions.

The system of equations for general equilibrium (15)-(17) must hold both before and after any counterfactual change in for example trade frictions. We denote the value of variables in the counterfactual equilibrium with a prime (x') and the relative value of variables in the counterfactual and initial equilibria by a tilde ($\tilde{x} = x'/x$). Using this notation, the system of equations for the counterfactual equilibrium (15)-(17) can be re-written as follows:

$$\tilde{w}_i \tilde{\lambda}_i Y_i = \sum_{n \in N} \pi'_{ni} \tilde{w}_n \tilde{\lambda}_n Y_n, \tag{19}$$

$$\pi_{ni}' = \frac{\pi_{ni} \left(\tilde{d}_{ni} \tilde{w}_i\right)^{-\theta}}{\sum_{k \in N} \pi_{nk} \left(\tilde{d}_{nk} \tilde{w}_k\right)^{-\theta}},\tag{20}$$

$$\tilde{\lambda}_{n} = \frac{\lambda_{n} \tilde{\pi}_{nn}^{-\frac{\alpha}{\theta(1-\alpha)}}}{\sum_{k \in N} \lambda_{k} \tilde{\pi}_{kk}^{-\frac{\alpha}{\theta(1-\alpha)}}},$$
(21)

where $Y_i = w_i L_i$ denotes labor income in the initial equilibrium.

This system of equations (19)-(21) can be solved for the counterfactual changes in wages (\tilde{w}_n) , population shares $(\tilde{\lambda}_n)$ and trade shares $(\tilde{\pi}_{ni})$ as a function of the exogenous change in trade costs \tilde{d}_{ni} and the values of observed variables in the initial equilibrium $\{Y_n, \pi_{ni}, \lambda_n\}$ for all locations $i, n \in N$. As discussed in the previous subsection, the change in each location's population share and its trade share with itself provide sufficient statistics for the change in its welfare between the initial and counterfactual equilibrium.

3 New Economic Geography

In our baseline model, the distribution of population across locations reflects a tension between differences in productivity (which concentrate population in high productivity locations), transport costs (which favor locations with good access to other locations) and an inelastic supply of land (which tends to disperse population). In this section, we show how our analysis can incorporate an endogenous agglomeration force in the form of pecuniary externalities from consumer love of variety, increasing returns to scale and transport costs, as in the new economic geography literature following Krugman (1991a,b). Specifically, we consider the Helpman (1998) model, which combines these endogenous agglomeration forces with a congestion force in the form of an inelastic supply of land. We show that a multi-region version of this model yields a system of equations for general equilibrium that takes exactly the same form as in our baseline model in the previous section. For parameter values for which there is a non-degenerate distribution of population across locations, this system of equations can be used to undertake model-based decompositions and counterfactuals in exactly the same way as in the previous section.

3.1 Consumer Preferences

Preferences are again defined over goods consumption (C_n) and residential land use (H_{Un}) and take the same form as in (1). The goods consumption index (C_n), however, is now defined over the endogenous measures of horizontally differentiated varieties supplied by each region (M_i):

$$C_{n} = \left[\sum_{i \in N} \int_{0}^{M_{i}} c_{ni} (j)^{\rho} dj\right]^{\frac{1}{\rho}},$$
(22)

where trade between regions *i* and *n* is again subject to iceberg variable trade costs of $d_{ni} \ge 1$.

3.2 Production

Varieties are produced under conditions of monopolistic competition. To produce a variety, a firm must incur a fixed cost of F units of labor and a constant variable cost in terms of labor that we

normalize to one. Therefore the total amount of labor $(l_n(j))$ required to produce $x_n(j)$ units of a variety *j* in country *n* is:

$$l_n(j) = F + x_n(j).$$
 (23)

Profit maximization and zero profits imply that equilibrium prices are a constant mark-up over marginal cost:

$$p_{ni}(j) = \left(\frac{\sigma}{\sigma - 1}\right) d_{ni} w_i,\tag{24}$$

and equilibrium output of each variety is equal to a constant:

$$x_n(j) = \bar{x} = F(\sigma - 1).$$
⁽²⁵⁾

Given constant equilibrium output of each variety, labor market clearing implies that the total measure of varieties supplied by each location is proportional to the endogenous supply of workers choosing to locate there:

$$M_i = \frac{L_i}{\sigma F}.$$
(26)

3.3 Price Indices and Expenditure Shares

Using equilibrium prices (24) and labor market clearing (26), the price index dual to the consumption index (22) can be expressed as:

$$P_n = \frac{\sigma}{\sigma - 1} \left(\frac{1}{\sigma F}\right)^{\frac{1}{1 - \sigma}} \left[\sum_{i \in N} L_i \left(d_{ni} w_i\right)^{1 - \sigma}\right]^{\frac{1}{1 - \sigma}}.$$
(27)

Using the CES expenditure function, equilibrium prices (24) and labor market clearing (26), the share of location n's expenditure on goods produced in location i is:

$$\pi_{ni} = \frac{M_i p_{ni}^{1-\sigma}}{\sum_{k \in N} M_k p_{nk}^{1-\sigma}} = \frac{L_i (d_{ni} w_i)^{1-\sigma}}{\sum_{k \in N} L_k (d_{nk} w_k)^{1-\sigma}}.$$
(28)

Together (27) and (28) imply that each location's price index can be again written in terms of its trade share with itself:

$$P_n = \frac{\sigma}{\sigma - 1} \left(\frac{L_n}{\sigma F \pi_{nn}} \right)^{\frac{1}{1 - \sigma}} w_n.$$
⁽²⁹⁾

3.4 Population Mobility and Welfare

As in the previous section, expenditure on land in each location is redistributed lump sum to the workers residing in that location, so that total income is proportional to labor income in (9). Population mobility again requires that workers receive the same real income in all populated locations as in (12). Using these two relationships together with land market clearing (11), welfare in each location can be expressed as follows:

$$V_n = \frac{\left(\frac{1}{\sigma F \pi_{nn}}\right)^{\frac{\alpha}{\sigma-1}} L_n^{-\left(\frac{\sigma(1-\alpha)-1}{\sigma-1}\right)} H_n^{1-\alpha}}{\alpha \left(\frac{1-\alpha}{\alpha}\right)^{1-\alpha} \left(\frac{\sigma}{\sigma-1}\right)^{\alpha}} = \bar{V},$$
(30)

where terms in wages (w_n) have again canceled and land quality (H_n) is a parameter.

Rearranging this population mobility condition, equilibrium population in each location can be expressed as a function of fundamentals (H_n), the location's trade share with itself (π_{nn}) and the common level of real income (\bar{V}) across locations:

$$L_{n} = \left[\frac{\left(\frac{1}{\sigma F \pi_{nn}}\right)^{\frac{\alpha}{\sigma-1}} H_{n}^{1-\alpha}}{\alpha \left(\frac{1-\alpha}{\alpha}\right)^{1-\alpha} \left(\frac{\sigma}{\sigma-1}\right)^{\alpha} \bar{V}}\right]^{\frac{\sigma-1}{\sigma(1-\alpha)-1}}.$$
(31)

A key difference between this model and the framework in the previous section is the presence of an endogenous agglomeration force as a result of the combination of love of variety, increasing returns to scale and transport costs. If this agglomeration force is sufficiently strong relative to the congestion force from an inelastic supply of land ($0 < \sigma(1 - \alpha) < 1$), each location's welfare (30) is increasing in its population for a given trade share with itself (π_{nn}), which generates the possibility of multiple equilibria and a degenerate population distribution with all population concentrated in a single region. Since we do not observe a degenerate population distribution in the data, we assume that the agglomeration force is sufficiently weak relative to the congestion force from an inelastic supply of land ($\sigma(1 - \alpha) > 1$), which ensures the existence of a stable equilibrium with a dispersed population distribution, as shown formally in the next subsection.

3.5 General Equilibrium

The general equilibrium of the model can be again represented by the share of workers in each location ($\lambda_n = L_n/\bar{L}$), the share of each location's expenditure on goods produced by other locations (π_{ni}) and the wage in each location (w_n). Using labor income (10), the trade share (28), population mobility (31) and labor market clearing (14), the equilibrium triple { λ_n , π_{ni} , w_n } solves the following system of equations for all $i, n \in N$:

$$w_i \lambda_i = \sum_{n \in N} \pi_{ni} w_n L_n, \tag{32}$$

$$\pi_{ni} = \frac{\lambda_n \left(d_{ni} w_i \right)^{1-\sigma}}{\sum_{k \in N} \lambda_k \left(d_{nk} w_k \right)^{1-\sigma}},\tag{33}$$

$$\lambda_{n} = \frac{\left[H_{n}^{1-\alpha}\left(\frac{1}{\pi_{nn}}\right)^{\frac{\alpha}{\sigma-1}}\right]^{\frac{\sigma-1}{\sigma(1-\alpha)-1}}}{\sum_{k \in N} \left[H_{k}^{1-\alpha}\left(\frac{1}{\pi_{kk}}\right)^{\frac{\alpha}{\sigma-1}}\right]^{\frac{\sigma-1}{\sigma(1-\alpha)-1}}}.$$
(34)

Proposition 4 Assuming $\sigma(1 - \alpha) > 1$ and given locational fundamentals (land quality (H_n) and bilateral trade frictions (d_{ni})), there exist equilibrium population shares (λ_n), trade shares (π_{ni}) and wages (w_n) that solve the wage equation (32), trade equation (33) and population mobility condition (34).

Proof. See the appendix.

While the above system of equations for general equilibrium takes the same form as in our baseline model in Section 2, relative population shares (λ_n) now enter the trade equation (33). The reason is consumer love of variety. Relative population shares determine the relative measures of varieties supplied by each location, and since all varieties are consumed, the relative measures of varieties supplied by each location influence trade shares.⁶

Since for $\sigma(1 - \alpha) > 1$ there exists a non-degenerate stable distribution of economic activity, the system of equations for general equilibrium (32)-(34) can be used to undertake model-based counterfactuals of the same form as in equations (19)-(21) in the previous section. Again changes in barriers to trade in goods result in endogenous changes in the distribution of population across locations, which now not only affect trade shares (33) indirectly through wages (w_n) but also directly through changes in the measures of varieties supplied by each location (captured by λ_n).

Given data on only population (L_n) and trade shares (π_{nn}), the population mobility condition (31) can be again used to measure unobserved locational fundamentals, which permits a model-based decomposition of the variance of population across locations into the contributions of locational fundamentals, market access and their covariance.

Proposition 5 Assuming $\sigma(1 - \alpha) > 1$ and given observed data on population (L_n) and trade shares (π_{ni}) , the population mobility condition (31) can be used to determine a measure of locational fundamentals in the form of land quality (H_n) . Using this measure, the variance of population across locations can be decomposed into the contributions of fundamentals $\left(H_n^{\frac{(1-\alpha)(\sigma-1)}{\sigma(1-\alpha)-1}}\right)$, market access $\left(\pi_{nn}^{\frac{-\alpha(\sigma-1)}{(\sigma-1)(\sigma(1-\alpha)-1)}}\right)$ and their covariance.

Proof. The proposition follows immediately from the population mobility condition (31) given observed data on population (L_n) and trade shares (π_{ni}):

$$\operatorname{var}\left\{\log L_{n}\right\} = \left(\frac{(1-\alpha)(\sigma-1)}{\sigma(1-\alpha)-1}\right)^{2}\operatorname{var}\left\{\log H_{n}\right\} + \left(\frac{\alpha(\sigma-1)}{(\sigma-1)(\sigma(1-\alpha)-1)}\right)^{2}\operatorname{var}\left\{\log \pi_{nn}\right\} \\ -2\left(\frac{(1-\alpha)(\sigma-1)}{\sigma(1-\alpha)-1}\right)\left(\frac{\alpha(\sigma-1)}{(\sigma-1)(\sigma(1-\alpha)-1)}\right)\operatorname{covar}\left\{\log H_{n},\log \pi_{nn}\right\}.$$

Population mobility again implies that the representative agent obtains the same welfare across all locations. As in our baseline model in the previous section, a location's trade share with itself is no longer a sufficient statistic for the representative agent's welfare gains from goods trade, which also depend on changes in the population distribution.

Proposition 6 *The change in a location's trade share with itself and the change in its population are sufficient statistics for the welfare gains from goods trade.*

⁶If the neoclassical model in the previous section were augmented to allow the productivity of each location to depend on its population through external economies of scale ($T_n = G_n L_n^{\eta}$ where $G_n, \eta > 0$), relative population shares would also enter the trade equation in that model. In that case, the existence of a non-degenerate population distribution would again impose a restriction on parameters that the agglomeration force created by these external economies of scale is not too strong relative to the congestion force from an inelastic supply of land.

Proof. The proposition follows immediately from (30) since:

$$\frac{V_n^{\text{Trade}}}{V_n^{\text{No Trade}}} = \left(\pi_{nn}^{\text{Trade}}\right)^{-\frac{\alpha}{\sigma-1}} \left(\frac{L_n^{\text{Trade}}}{L_n^{\text{No Trade}}}\right)^{-\left(\frac{\sigma(1-\alpha)-1}{\sigma-1}\right)} = \frac{\bar{V}^{\text{Trade}}}{\bar{V}^{\text{No Trade}}},$$
(35)

where $\pi_{nn}^{\text{No Trade}} = 1$ and $\sigma(1 - \alpha) > 1$ is required for a non-degenerate population distribution.

Therefore our result that reallocations of mobile factors of production provide a new channel through which goods trade affects welfare is not specific to the neoclassical model of the previous section, but also holds in other models within the class considered by Arkolakis et al. (2012) when augmented to include mobile and immobile factors. While we demonstrate this result in this section using a model of love of variety preferences, the same result holds in a model of national product differentiation following Armington (1969).

4 Imperfect Labor Mobility

To illustrate the effects of factor mobility on the impact of goods trade liberalization, we have so far considered the case in which the mobile factor (labor) is perfectly mobile across locations. In this section, we extend the analysis to allow for imperfect labor mobility. We capture imperfect labor mobility by assuming that workers have idiosyncratic preferences for each location. But these idiosyncratic preferences also have an isomorphic interpretation as idiosyncratic mobility costs or idiosyncratic worker productivities for each location.

Labor mobility now implies that each worker chooses the region that offers her the highest utility (after taking into account her idiosyncratic preferences). In general, real wages now differ across locations. As the distribution of idiosyncratic preferences becomes degenerate, we obtain our baseline version of the model, in which labor mobility ensures that real wages are equalized across locations. More generally, with a non-degenerate distribution of idiosyncratic preferences, each region faces an upward-sloping supply curve for workers, because higher real wages have to be paid to attract workers with lower tastes for a region.

We return to our baseline model from Section 2. Consumer preferences take the same form as in (1), except that utility now also depends on idiosyncratic amenity draws for each location:

$$U_n(\omega) = b_n(\omega) \left(\frac{C_n}{\alpha}\right)^{\alpha} \left(\frac{H_{Un}}{1-\alpha}\right)^{1-\alpha}, \qquad 0 < \alpha < 1.$$
(36)

Each worker ω draws an idiosyncratic amenity draw $b_n(\omega)$ for each location n. Amenities are independently drawn across workers and locations from a Fréchet distribution:

$$G_n(b) = e^{-B_n b^{-\epsilon}},\tag{37}$$

where the scale parameter B_n determines average amenities for location n and the shape parameter ϵ controls the dispersion of amenities across workers for each location. The corresponding indirect

utility function is:

$$U_n(\omega) = \frac{b_n(\omega)v_n}{P_n^{\alpha}r_n^{1-\alpha}}.$$
(38)

Since indirect utility is a monotonic function of the amenity draw, it too has a Fréchet distribution:

$$G_n(U) = e^{-\psi_n U^{-\epsilon}}, \qquad \psi_n = B_n \left(P_n^{\alpha} r_n^{1-\alpha} \right)^{-\epsilon} v_n^{\epsilon}.$$

Therefore the probability that a worker chooses to live in region *n* is:

$$\lambda_n = \frac{L_n}{\bar{L}} = \frac{B_n \left(P_n^{\alpha} r_n^{1-\alpha} \right)^{-\epsilon} v_n^{\epsilon}}{\sum_{k \in N} B_k \left(P_k^{\alpha} r_k^{1-\alpha} \right)^{-\epsilon} v_k^{\epsilon}}.$$
(39)

Using land market clearing (11), the equality of income and expenditure (9) and the price index (8), the above residential choice probability can be re-written as:

$$\lambda_n = \frac{B_n \left(\frac{T_n}{\pi_{nn}}\right)^{\frac{\alpha\varepsilon}{\theta}} \left(\frac{L_n}{H_n}\right)^{-\epsilon(1-\alpha)}}{\sum_{k \in N} B_k \left(\frac{T_k}{\pi_{kk}}\right)^{\frac{\alpha\varepsilon}{\theta}} \left(\frac{L_k}{H_k}\right)^{-\epsilon(1-\alpha)}},$$
(40)

which corresponds to a generalization of (13) to incorporate imperfect labor mobility, and implicitly determines equilibrium population shares as a function of locational fundamentals (productivity (T_n) and land quality (H_n)), and each location's trade share with itself (π_{nn}).

The general equilibrium of the model again can be represented by the share of workers in each location ($\lambda_n = L_n/\bar{L}$), the share of each location's expenditure on goods produced in other locations (π_{ni}) and the wage in each location (w_n). Using labor income (10), the trade share (6), and the residential choice probability (40), this equilibrium triple { λ_n, π_{ni}, w_n } solves the following system of equations for all $i, n \in N$:

$$w_i \lambda_i = \sum_{n \in N} \pi_{ni} w_n \lambda_n, \tag{41}$$

$$\pi_{ni} = \frac{T_i \left(d_{ni} w_i \right)^{-\theta}}{\sum_{k \in N} T_k \left(d_{nk} w_k \right)^{-\theta}},\tag{42}$$

$$\lambda_n = \frac{B_n \left(\frac{T_n}{\pi_{nn}}\right)^{\frac{\alpha\epsilon}{\theta}} \left(\frac{L_n}{H_n}\right)^{-\epsilon(1-\alpha)}}{\sum_{k \in N} B_k \left(\frac{T_k}{\pi_{kk}}\right)^{\frac{\alpha\epsilon}{\theta}} \left(\frac{L_k}{H_k}\right)^{-\epsilon(1-\alpha)}}.$$
(43)

Therefore imperfect labor mobility can be incorporated into this class of quantitative models and handled in a similar way as perfect labor mobility.

5 Internal Versus External Trade

To simplify the exposition of our argument, we have so far considered contexts in which the mobile factor of production is mobile across all locations. In this section, we consider a more realistic setting of an economy containing distinct countries each of which consists of many regions. Labor is perfectly mobile across regions within countries but immobile between countries. The resulting extension of our baseline model from Section 2 provides a natural platform for examining the role of reallocations of resources across regions within countries in shaping the welfare gains from reductions in trade costs between countries. With a view to taking the model to the data, we also generalize our previous analysis to incorporate commercial land use and traded intermediate inputs. We show that the augmented model yields an analogous system of equations for general equilibrium that can be again used to undertake model-based decompositions and counterfactuals.

5.1 Consumer Preferences and Production

The world economy consists of a set N of regions indexed by n. A subset $N^A \subset N$ of these regions are in a home country and a subset $N^B \subset N$ are in a foreign country, where $N^A \cap N^B = \emptyset$. While for simplicity we focus on the case of two countries, it is straightforward to accommodate an arbitrary number of countries that each consist of many regions.

Preferences are again defined over goods consumption (C_n) and residential land use (H_{Un}) and take the same form as in (1). The goods consumption index (C_n) is defined over consumption of a fixed continuum of goods $j \in [0, 1]$ as in (2).

Each region draws an idiosyncratic productivity z_j for each good j as in (3) and goods are again homogeneous in the sense that one unit of a given good is the same as any other unit of that good. Goods are produced with labor, land and intermediate inputs under conditions of perfect competition according to a Cobb-Douglas production technology. The cost to a consumer in region n of purchasing one unit of good j from region i is therefore:

$$p_{ni}(j) = \frac{d_{ni}w_i^{\beta}r_i^{\eta}P_i^{1-\beta-\eta}}{z_i(j)}, \qquad 0 < \beta < 1, \ 0 < \eta < 1,$$
(44)

where P_i is the price index for tradable goods.

5.2 Expenditure Shares and Price Indices

The representative consumer in region n sources each good from the lowest cost supplier to that region. Using equilibrium prices (44) and the properties of the Fréchet distribution, the share of expenditure of region n on goods produced by region i is:

$$\pi_{ni} = \frac{T_i \left(d_{ni} w_i^{\beta} r_i^{\eta} P_i^{1-\beta-\eta} \right)^{-\theta}}{\sum_{j \in \{A,B\}} \sum_{k \in N^j} T_k \left(d_{nk} w_k^{\beta} r_k^{\eta} P_k^{1-\beta-\eta} \right)^{-\theta}},$$
(45)

while the price index for tradable goods is:

$$P_n = \gamma \left[\sum_{j \in \{A,B\}} \sum_{i \in N^j} T_i \left(d_{ni} w_i^\beta r_i^\eta P_i^{1-\beta-\eta} \right)^{-\theta} \right]^{1/\theta},$$
(46)

where for clarity we separate out the regions located in each country in the summations.

5.3 Income and Population Mobility

Expenditure on land in each region is again redistributed lump sum to the workers residing in that region. Total income in each region is therefore:

$$v_n L_n = w_n L_n + (1 - \alpha) v_n L_n + \eta R_n,$$

where R_n denotes the total revenue from production; the first term on the right-hand side is income from labor used in production; the second term is income from expenditure on residential land; and the third term is income from expenditure on commercial land.

Noting that payments to labor used in production are a constant share of revenue, $w_n L_n = \beta R_n$, total income in each region can be re-written as:

$$v_n L_n = \left(\frac{\beta + \eta}{\alpha\beta}\right) w_n L_n. \tag{47}$$

Goods market clearing requires that total revenue in each region equals expenditure on goods produced in that region:

$$R_i = \sum_{j \in \{A,B\}} \sum_{n \in N^j} \pi_{ni} \left[\alpha v_n L_n + (1 - \beta - \eta) R_n \right],$$

where the first term inside the square parentheses is expenditure from final goods consumption and the second term is expenditure from intermediate inputs use. Using total income (47) and $w_n L_n = \beta R_n$, this goods market clearing condition can be written in exactly the same form as (10).

Land market clearing implies that the equilibrium land rent can be determined from the equality of land income and expenditure:

$$r_n = \frac{(1-\alpha)v_nL_n + \eta R_n}{H_n} = \left(\frac{(1-\alpha)\beta + \eta}{\alpha\beta}\right)\frac{w_nL_n}{H_n},\tag{48}$$

where we have again used $w_n L_n = \beta R_n$.

Intra-national population mobility ensures that real incomes are equalized across regions within each country, while international population immobility implies that real incomes can differ between countries. Using land market clearing (48), the equality of income and expenditure (47), the expenditure share (45) and the price index (46) in the population mobility condition (12), welfare in each location within a given country can be expressed as follows:

$$V_{n} = \frac{\left(\frac{T_{n}}{\pi_{nn}}\right)^{\frac{\alpha}{\theta(\beta+\eta)}} L_{n}^{-\frac{\eta+(1-\alpha)\beta}{\beta+\eta}} H_{n}^{\frac{\eta+(1-\alpha)\beta}{\beta+\eta}}}{\gamma^{\frac{\alpha}{\beta+\eta}} \frac{\alpha\beta}{\beta+\eta} \left(\frac{\eta+(1-\alpha)\beta}{\alpha\beta}\right)^{\frac{\eta+(1-\alpha)\beta}{\beta+\eta}}} = \bar{V}^{j}, \qquad n \in N^{j}, \ j \in \{A, B\},$$

$$(49)$$

where terms in wages (w_n) have again canceled; land quality (H_n) is a parameter; and \bar{V}^j is the common level of utility across regions within country *j*.

Rearranging this population mobility condition, equilibrium population in each region within a country can be expressed in terms of fundamentals (productivity (T_n) and land quality (H_n)), the

region's trade share with itself (π_{nn}) and the common level of real income across regions within that country (\bar{V}^j for $j \in \{A, B\}$):

$$L_{n} = \frac{\left(\frac{T_{n}}{\pi_{nn}}\right)^{\frac{\alpha}{\theta(\eta+(1-\alpha)\beta)}} H_{n}}{\left(\frac{\alpha\beta}{\beta+\eta}\right)^{\frac{\beta+\eta}{\eta+(1-\alpha)\beta}} \left(\frac{\eta+(1-\alpha)\beta}{\alpha\beta}\right) \gamma^{\frac{\alpha}{\eta+(1-\alpha)\beta}} (\bar{V}^{j})^{\frac{\beta+\eta}{\eta+(1-\alpha)\beta}}, \qquad n \in N^{j}, \ j \in \{A, B\},$$
(50)

where the labor market clearing condition (14) holds for each country separately.

5.4 General Equilibrium

The general equilibrium of the model can be represented by the share of each country's workers located in each region ($\lambda_n^j = L_n/\bar{L}^j$ for $j \in \{A, B\}$), the share of each location's expenditure on goods produced in other locations (π_{ni}), the wage in each location (w_n), and the price index in each location (P_n). Using labor income (47), the trade share (45), the price index (46), land market clearing (48), labor market clearing (14) and population mobility (50), the equilibrium quintuple { $\lambda_n^A, \lambda_n^B, \pi_{ni}, w_n, P_n$ } solves the following system of equations for the two countries:

$$w_i \lambda_i^s \bar{L}^s = \sum_{j \in \{A,B\}} \sum_{n \in N^j} \pi_{ni} w_n \lambda_n^j \bar{L}^j, \quad i \in N^s, \, s \in \{A,B\}$$
(51)

$$\pi_{ni} = \frac{T_i \left(d_{ni} w_i^{\beta+\eta} P_i^{1-\beta-\eta} \left(\lambda_i^s \right)^\eta \left(\frac{\bar{L}^s}{\bar{H}_i} \right)^\eta \right)^{-\theta}}{\sum_{i \in \{A,B\}} \sum_{k \in N^i} T_k \left(d_{nk} w_k^{\beta+\eta} P_k^{1-\beta-\eta} \left(\lambda_k^j \right)^\eta \left(\frac{\bar{L}^j}{\bar{H}_k} \right)^\eta \right)^{-\theta}}, \quad i \in N^s, \, s \in \{A, B\},$$
(52)

$$P_{n} = \gamma \left[\sum_{j \in \{A,B\}} \sum_{k \in N^{j}} T_{k} \left(\kappa_{P} d_{nk} w_{k}^{\beta+\eta} P_{k}^{1-\beta-\eta} \left(\lambda_{k}^{j} \right)^{\eta} \left(\frac{\bar{L}^{j}}{H_{k}} \right)^{\eta} \right)^{-\theta} \right]^{-1/\theta}, \quad n \in N^{s}, s \in \{A,B\},$$
(53)

$$\lambda_n^s = \frac{\left(\frac{T_n}{\pi_{nn}}\right)^{\theta(\beta+\eta-\alpha\beta)} H_n}{\sum_{k \in N^s} \left(\frac{T_k}{\pi_{kk}}\right)^{\frac{\alpha}{\theta(\beta+\eta-\alpha\beta)}} H_k}, \qquad n \in N^s, \, s \in \{A, B\}$$
(54)

where \bar{L}^s is the exogenous endowment of labor for country *s* and H_n is the exogenous endowment of land for each region *n* and

$$\kappa_P \equiv \left(\frac{(1-\alpha)\beta + \eta}{\alpha\beta}\right)^{\eta}.$$
(55)

Proposition 7 Given locational fundamentals (productivity (T_n) , land quality (H_n) and bilateral trade frictions (d_{ni})), there exist equilibrium population shares (λ_n) , trade shares (π_{ni}) , wages (w_n) and price indices (P_n) that solve the wage equation (51), trade equation (52), price equation (53) and population mobility condition (54).

Proof. See the appendix.

While the wage equation (51), trade equation (52) and price index (53) take a similar form as in previous sections, there are also some differences. First, the population mobility condition (54) now

only holds within each country. Second, population shares (λ_n) now enter the trade equation (52), because land is used in production and a higher population implies a higher price for immobile land and hence higher production costs. Third, in the previous two sections, equilibrium wages, trade shares and population shares { w_n , π_{ni} , λ_n } could be determined before price indices (P_n), because the absence of intermediate inputs from production gave rise to a recursive structure of the model. In contrast, in this section, price indices (P_n) enter the trade equation because of intermediate input use in production. As a result, price indices (P_n) must be jointly determined with wages (w_n), trade shares (λ_n) and population shares (λ_n) in the system of equations for general equilibrium.

In this more general setting, the population mobility condition (50) together with observed data on population and trade shares can be still used to decompose the variance of population across locations into the contributions of locational fundamentals and market access.

Proposition 8 Given observed data on population (L_n) and trade shares (π_{ni}) , the population mobility condition (50) can be used to determine a composite measure of fundamentals for each location that incorporates productivity and land quality $\left(H_n T_n^{\frac{\alpha}{\theta(\beta+\eta-\alpha\beta)}}\right)$. Using this measure, the variance of population across locations can be decomposed into the contributions of fundamentals $\left(H_n T_n^{\frac{\alpha}{\theta(\beta+\eta-\alpha\beta)}}\right)$, market access $\left(\pi_{nn}^{\frac{-\alpha}{\theta(\beta+\eta-\alpha\beta)}}\right)$ and their covariance.

Proof. The proposition follows immediately from the population mobility condition (50) given observed data on population (L_n) and trade shares (π_{ni}):

$$\operatorname{var}\left\{\log L_{n}\right\} = \operatorname{var}\left\{\log\left(T_{n}^{\alpha/\theta(\beta+\eta-\alpha\beta)}H_{n}\right)\right\} + \left(\frac{\alpha}{\theta(\beta+\eta-\alpha\beta)}\right)^{2}\operatorname{var}\left\{\log\pi_{nn}\right\} \\ -2\left(\frac{\alpha}{\theta(\beta+\eta-\alpha\beta)}\right)\operatorname{covar}\left\{\log\left(T_{n}^{\alpha/\theta(\beta+\eta-\alpha\beta)}H_{n}\right),\log\left(\pi_{nn}\right)\right\}\right\}$$

The system of equations for general equilibrium (51)-(54) can be used to undertake model-based counterfactuals of the same form as in equations (19)-(21) in the previous section. These counterfactuals now yield predictions for the internal distribution of economic activity across regions within countries. As in our baseline model, a region's trade share with itself is no longer a sufficient statistic for the welfare gains from market integration, which also depend on population changes.

Proposition 9 *The change in a location's trade share with itself and the change in its population are sufficient statistics for the welfare gains from goods trade.*

Proof. The proposition follows immediately from (49) since:

$$\frac{V_n^{\text{Trade}}}{V_n^{\text{No Trade}}} = \left(\pi_{nn}^{\text{Trade}}\right)^{-\frac{\alpha}{\theta(\beta+\eta)}} \left(\frac{L_n^{\text{Trade}}}{L_n^{\text{No Trade}}}\right)^{-\left(1-\frac{\alpha\beta}{\beta+\eta}\right)},$$
(56)

where $\pi_{nn}^{\text{No Trade}} = 1.$

Therefore the effect of changes in trade costs between countries on regional welfare depends on whether or not population is mobile within countries. If population is geographically immobile, the change in each region's trade share with itself is a sufficient statistic for the welfare effects of international goods market integration. Furthermore, border regions that have higher trade shares with the other country in the open economy equilibrium experience greater welfare gains from international goods market integration than interior regions that have lower trade shares with the other country. In contrast, if population is mobile across regions within countries, the change in each region's welfare also depends on the change in its population, which affects welfare through the price of land. External trade liberalization leads to an endogenous internal reallocation of population such that all regions experience the same welfare gains from international goods market integration.

Thus endogenous changes in the distribution of mobile factors across regions within countries affect the representative agent's welfare gains from trade liberalization between countries. As a result, assuming that each country consists of a single region and applying the standard welfare gains from trade formula (Arkolakis et al. 2012) yields a biased estimate of the representative agent's welfare gains from trade if each country in reality consists of multiple regions linked by factor mobility. The model implies a bias correction term for adjusting the standard welfare gains from trade formula to take account of endogenous reallocations of resources across regions within countries in response to trade liberalization. To derive this bias correction term, note that the expression for welfare (49) would hold in a hypothetical world in which each country consisted of a single region:

$$\tilde{V}_{j} = \frac{\left(\frac{T_{j}}{\tilde{\pi}_{jj}}\right)^{\frac{\alpha}{\theta(\beta+\eta)}} \tilde{L}_{j}^{-\frac{\eta+(1-\alpha)\beta}{\beta+\eta}} \tilde{H}_{j}^{\frac{\eta+(1-\alpha)\beta}{\beta+\eta}}}{\gamma^{\frac{\alpha}{\beta+\eta}} \frac{\alpha\beta}{\beta+\eta} \left(\frac{\eta+(1-\alpha)\beta}{\alpha\beta}\right)^{\frac{\eta+(1-\alpha)\beta}{\beta+\eta}}}, \qquad j \in \{A, B\},$$
(57)

where a tilde denotes the value of a variable in the hypothetical world in which each country $j \in \{A, B\}$ consists of a single region. Therefore \tilde{L}_j and \tilde{H}_j correspond to the exogenous aggregate population and land supply of country j.

Comparing welfare (49) in the actual world in which each country consists of many regions linked by labor mobility with welfare (57) in the hypothetical world, the actual welfare gains from trade equal a bias correction term times the hypothetical welfare gains from trade:

$$\frac{V_n^{\text{Trade}}}{V_n^{\text{No Trade}}} = \left(\frac{\tilde{\pi}_{jj}^{\text{Trade}}}{\pi_{nn}^{\text{Trade}}}\right)^{\frac{\alpha}{\theta(\beta+\eta)}} \left(\frac{L_n^{\text{Trade}}}{L_n^{\text{No Trade}}}\right)^{-\left(1-\frac{\alpha\beta}{\beta+\eta}\right)} \frac{\tilde{V}_n^{\text{Trade}}}{\tilde{V}_n^{\text{No Trade}}},\tag{58}$$

where the hypothetical welfare gains from trade are:

$$\frac{\tilde{V}_{n}^{\text{Trade}}}{\tilde{V}_{n}^{\text{No Trade}}} = \left(\tilde{\pi}_{jj}^{\text{Trade}}\right)^{-\frac{\alpha}{\theta(\beta+\eta)}}.$$
(59)

The bias correction term in (58) controls for (i) the difference between the open economy domestic trade share of an individual region within a country and the aggregate domestic trade share of the country as a whole and (ii) the change in distribution of the mobile factor of production across regions within the country in response to trade liberalization.

6 Quantitative Application

In this section, we use the model developed in the previous section together with data on trade between Canadian and U.S. regions to illustrate the quantitative relevance of factor mobility within countries for the welfare effects of goods trade. We begin by discussing the data before using the structure of the model to evaluate the welfare implications of counterfactual changes in barriers to goods trade. We examine how these welfare implications depend on whether population is mobile or immobile across regions within each country.

6.1 Data Sources and Definitions

Our main source of data is the information on trade between Canadian provinces and U.S. states for 1993 from Anderson and Van Wincoop (2003). A key feature of these data is that they contain information on internal trade between regions within each country and external trade between regions in the two different countries. We combine these data with information on the population of Canadian provinces and U.S. states from the closest Census year, which is 1990.

To take the model to the data, a number of choices must be made. To simplify the exposition of the model, we assumed a continuum of products, which implies that each location trades with all other locations.⁷ To ensure that the data are consistent with this feature of the model, we aggregate Canadian provinces and U.S. states to the level of statistical regions.⁸ For expositional convenience, we also focused in the model on two countries and assumed that trade is balanced (income equals expenditure). Since the goal of our empirical analysis is merely to illustrate the implications of factor mobility for the welfare gains from goods trade, we make analogous assumptions in the data. We use actual bilateral trade flows between Canadian and U.S. regions to construct a matrix of bilateral import shares for each region that we use as an input into the model. In using this matrix of bilateral is balanced for each region separately and hence for the two countries as a whole. Introducing the rest of the world as a third country and allowing for trade imbalance is straightforward, and would alter the specific numerical predictions under population immobility and mobility, but it would not change the basic contrast between these two cases that we illustrate in this section.

In Table 1, we report the matrix of bilateral import shares for Canadian and U.S. regions in 1993, where the importer appears in the columns and the exporter appears in the rows, such that the rows sum to one for each column. Unsurprisingly, the largest share of a region's trade is with itself. But these own trade shares for Canadian and U.S. regions are smaller than for Canada and the U.S. as a whole, which reflects the regions' smaller economic size and implies greater welfare gains from trade with other regions. Within Canada, the largest trade flows between pairs of distinct regions

⁷In principle, the model in the previous section can be used to consider a finite number of products. In this case, each region draws a finite number of productivities, for which some pairs of regions may not trade, as in Eaton et al. (2012).

⁸See the data appendix for the definition of statistical regions. To ensure positive bilateral trade flows between all pairs of regions, we exclude Northern Canada, which accounts for a small share of Canadian trade and population.

are imports of Central Canada from Atlantic Canada and imports of Western Canada from Central Canada. Within the U.S., the largest trade flows between pairs of distinct regions are imports of the South from the Mid West and imports of the North East from the South. Between the two countries, the most intense trade relationship by far is imports of Central Canada from the Mid West, followed by imports of the other Canadian regions from the Mid West, and by imports of the North East from Central Canada. Reflecting the larger economic size of the U.S. relative to Canada, U.S. regions together account for 15-25 percent of Canadian regions' imports, whereas Canadian regions together account for 5 percent or less of U.S. regions' imports.

		Importer						
		Atl. Can.	Cent. Can.	West Can.	Mid West	North East	South	West
Exporter	Atl. Can.	0.6006	0.0093	0.0044	0.0009	0.0044	0.0008	0.0001
	Cent. Can.	0.2161	0.6966	0.1205	0.0385	0.0395	0.0070	0.0027
	West. Can.	0.0266	0.0385	0.7305	0.0097	0.0058	0.0026	0.0041
	All Can.	0.8434	0.7445	0.8554	0.0490	0.0497	0.0105	0.0069
Exporter	Mid West	0.0574	0.1322	0.0547	0.4656	0.1441	0.3082	0.2046
	North East	0.0465	0.0550	0.0181	0.1263	0.3174	0.2006	0.1285
	South	0.0392	0.0503	0.0308	0.2948	0.2972	0.3681	0.3641
	West	0.0136	0.0180	0.0410	0.0643	0.1916	0.1127	0.2960
	All U.S.	0.1566	0.2555	0.1446	0.9510	0.9503	0.9895	0.9931
	All regions	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000	1.0000

Table 1: Import Shares of U.S. and Canadian Regions

Note: Data source is Anderson and van Wincoop (2003). See the appendix for definitions of U.S. and Canadian regions. Importers appear in columns and exporters appear in rows; sub-totals for all Canadian regions and all U.S. regions sum to the total for all regions; figures need not exactly sum to one due to rounding.

In Table 2, we report the share of each region in the population sum for the country as a whole. Central Canada accounts for by far the largest share of the Canadian population, while the South is the most populous U.S. region. In general, population is more equally distributed across U.S. regions than across Canadian regions, with Atlantic Canada having the smallest population share of all regions. Taking the import and population shares reported in Tables 1 and 2 together with the population sum for Canada and the U.S., we use the wage equation (51) to solve for the implied wages (w_i) and hence labor income ($Y_i = w_i \lambda_i^s \bar{L}^s$) of each region *i* in the initial equilibrium of the model prior to the counterfactual change in trade costs considered below.

The import shares (π_{ni}), population shares (λ_n) and labor income (Y_n) in the initial equilibrium are inputs into our counterfactual analysis. For the model's parameters, we assume central values from the existing literature. We set the share of residential land in consumer expenditure ($1 - \alpha$) equal to 0.25, which corresponds to the estimated housing expenditure share using U.S. data in Davis and Ortalo-Magne (2011). We set the share of intermediate inputs in production costs ($1 - \beta - \eta$) equal to 0.50, which is consistent with empirical findings for the manufacturing sector. We set the share of land in production costs (η) equal to 0.10 and the share of labor in production costs (β) equal to

1			0	
Canada	Population	U.S.	Population	
	Shares		Shares	
Atlantic Canada	0.0854	Mid West	0.2400	
Central Canada	0.6265	North East	0.2043	
Western Canada	0.2881	South	0.3436	
		West	0.2121	
Total	1.0000	Total	1.0000	

Table 2: Population Shares of U.S. and Canadian Regions

Note: Data sources are the U.S. Census Bureau and Statistics Canada. See the appendix for definitions of U.S. and Canadian regions. The population sum for Canada excludes Northern Canada. Population shares sum to one for each country separately up to rounding.

0.40. These values imply a share of immobile factors in value added of around 20 percent, which is in line with the estimates in Valentinyi and Herrendorf (2008). Finally, we set the Fréchet shape parameter (θ) equal to 4, which lies in the centre of the range of estimates in Donaldson (2014), Eaton and Kortum (2002) and Simonovska and Waugh (2013).

6.2 Counterfactuals under Population Immobility

To illustrate the implications of population mobility for the welfare gains from goods trade, we consider a change in bilateral trade frictions that reduces international trade costs between Canadian and U.S. regions by 25 percent while leaving intra-national trade costs between regions within each country unchanged. In this subsection, we solve for the counterfactual implications of this change in trade costs for the special case of the model in which labor is immobile across regions. In this case, wages (w_n), trade shares (π_{ni}) and price indices (P_n) are determined by the wage equation (51), trade equation (52) and price index equation (53), while population shares are exogenously determined by endowments. Denoting the value of variables in the initial equilibrium by a prime (x') and the relative value of variables in the counterfactual and initial equilibria by a tilde ($\tilde{x} = x'/x$), the wage, trade and price index equations in the counterfactual equilibrium can be re-written as follows:

$$\tilde{w}_i Y_i = \sum_{j \in \{A,B\}} \sum_{n \in N^j} \pi'_{ni} \tilde{w}_n Y_n, \qquad i \in N^j, \ j \in \{A,B\}$$
(60)

$$\pi_{ni}' = \frac{\pi_{ni} \left(\tilde{d}_{ni} \tilde{w}_i^{\beta+\eta} \tilde{P}_i^{1-\beta-\eta} \right)^{-\theta}}{\sum_{j \in \{A,B\}} \sum_{k \in N^j} \pi_{nk} \left(\tilde{d}_{nk} \tilde{w}_k^{\beta+\eta} \tilde{P}_k^{1-\beta-\eta} \right)^{-\theta}}, \quad i, n \in N,$$
(61)

$$\tilde{P}_n = \gamma \left[\sum_{j \in \{A,B\}} \sum_{k \in N^j} \pi_{nk} \left(\tilde{d}_{nk} \tilde{w}_k^{\beta+\eta} \tilde{P}_k^{1-\beta-\eta} \right)^{-\theta} \right]^{-1/\theta}, \quad n \in N,$$
(62)

where $Y_i = w_i \lambda_i^s \bar{L}^s$ denotes labor income in the initial equilibrium; the comparative static considered in the counterfactual corresponds to $\tilde{d}_{ni} = 1$ when regions *n* and *i* are located in the same country and $\tilde{d}_{ni} = 0.75$ when regions *n* and *i* are located in different countries. Together (60)-(62) provide a system of equations for the *N* locations that can be solved for the counterfactual changes in wages, trade shares and price indices $\{\tilde{w}_n, \tilde{\pi}_{ni}, \tilde{P}_n\}$ for each region. Given these solutions, the counterfactual change in welfare for each region under factor immobility follows immediately from the change in each region's trade share with itself:

$$\tilde{V}_n = \frac{V'_n}{V_n} = \left(\frac{\pi'_{nn}}{\pi_{nn}}\right)^{-\frac{\alpha}{\theta(\beta+\eta)}} = (\tilde{\pi}_{nn})^{-\frac{\alpha}{\theta(\beta+\eta)}}.$$
(63)

In Table 3, we report the counterfactual change in own trade shares and welfare for each region. Consistent with imports from the U.S. being more important for Canadian regions than imports from Canada are for U.S. regions, the counterfactual changes in own trade shares and welfare are larger for Canadian regions than U.S. regions. Across Canadian regions, Central Canada benefits by more than Western Canada, because of its greater trade intensity with U.S. regions including in particular the Mid West. Across U.S. regions, the Mid West and North East benefit by more than the South and West because of their greater trade integration with Canada.

Region	Relative Own	Relative
C	Trade Shares	Welfare
	π'_{nn}/π_{nn}	V_n'/V_n
Atlantic Canada	0.6833	1.1535
Central Canada	0.5995	1.2115
Western Canada	0.7340	1.1230
Mid West	0.9130	1.0347
North East	0.9241	1.0301
South	0.9878	1.0046
West	0.9848	1.0057

 Table 3: Counterfactuals for Welfare under Population Immobility

Note: The table reports the results from a counterfactual in which international trade costs between Canadian and U.S. regions are reduced by 25 percent while intra-national trade costs between regions within each country remain unchanged; variables in the counterfactual equilibrium are denoted by a prime; variables in the initial equilibrium are denoted without a prime. See the appendix for definitions of U.S. and Canadian regions.

6.3 Counterfactuals under Population Mobility

We now examine the counterfactual implications of the same change in barriers to goods trade under population mobility. Wages (w_n), trade shares (π_{ni}), price indices (P_n) and population shares (λ_n) are determined by the wage equation (51), trade equation (52), price index equation (53) and population mobility condition (54). Using the same notation as above, this system of equations in the counterfactual equilibrium can be re-written as follows:

$$\tilde{w}_i \tilde{\lambda}_i^s Y_i = \sum_{j \in \{A,B\}} \sum_{n \in N^j} \pi'_{ni} \tilde{w}_n \tilde{\lambda}_n^j Y_n, \qquad i \in N^j, \, s, j \in \{A,B\}$$
(64)

$$\pi_{ni}' = \frac{\pi_{ni} \left(\tilde{d}_{ni} \tilde{w}_i^{\beta+\eta} \tilde{P}_i^{1-\beta-\eta} \left(\tilde{\lambda}_i^A \right)^\eta \right)^{-\theta}}{\sum_{j \in \{A,B\}} \sum_{k \in N^j} \pi_{nk} \left(\tilde{d}_{nk} \tilde{w}_k^{\beta+\eta} \tilde{P}_k^{1-\beta-\eta} \left(\tilde{\lambda}_k^j \right)^\eta \right)^{-\theta}}, \quad i, n \in N,$$
(65)

$$\tilde{P}_{n} = \gamma \left[\sum_{j \in \{A,B\}} \sum_{k \in N^{j}} \pi_{nk} \left(\tilde{d}_{nk} \tilde{w}_{k}^{\beta+\eta} \tilde{P}_{k}^{1-\beta-\eta} \left(\tilde{\lambda}_{k}^{j} \right)^{\eta} \right)^{-\theta} \right]^{-1/\theta}, \quad n \in N,$$
(66)

$$\lambda_n^{s\prime} = \frac{\lambda_n^s \tilde{\pi}_{nn}^{-\frac{\alpha}{\theta(\beta+\eta-\alpha\beta)}}}{\sum_{k \in N^s} \lambda_k^s \tilde{\pi}_{kk}^{-\frac{\alpha}{\theta(\beta+\eta-\alpha\beta)}}}, \qquad n \in N^s, \, s \in \{A, B\},\tag{67}$$

where $Y_i = w_i \lambda_i^s \bar{L}^s$ again denotes labor income in the initial equilibrium; the comparative static considered in the counterfactual again corresponds to $\tilde{d}_{ni} = 1$ when regions *n* and *i* are located in the same country and $\tilde{d}_{ni} = 0.75$ when regions *n* and *i* are located in different countries.

The general equilibrium system (64)-(67) can be solved for the counterfactual changes in wages, trade shares, price indices and population shares $\{\tilde{w}_n, \tilde{\pi}_{ni}, \tilde{P}_n, \tilde{\lambda}_n^s\}$ for each region $n \in N^s$ for each country *s*. Given these solutions, the counterfactual change in each region's welfare under population mobility follows immediately from (56) and depends on both the change in each region's trade share with itself and the change in its population.

In Table 4, we report the counterfactual changes in own trade shares, welfare and population shares for each region. Consistent with imports from the U.S. being more important for Canadian regions than imports from Canada are for U.S. regions, the counterfactual changes in own trade shares and welfare are again larger for Canadian regions than U.S. regions. In the results in the previous subsection under population immobility, the counterfactual changes in welfare differed across regions within each country depending on their initial level of trade integration with regions in the other country. In contrast under population mobility, internal reallocations of population ensure that the external trade liberalization has the same effect on welfare in each region, raising welfare by around 18 percent for Canadian regions and by around 2 percent for U.S. regions.

Central Canada's higher initial import shares from the U.S. than other Canadian regions imply that the reduction in bilateral trade frictions reduces the consumer price index in Central Canada relative to the other Canadian regions. As a result, Central Canada becomes a more attractive place to live, which induces a population inflow from the other Canadian regions. The mechanisms that restore equilibrium are changes in wages and the price of land across regions. As the population of Central Canada rises relative to other Canadian regions, this bids up the price of land in Central Canada relative to other Canadian regions, until real wages are equalized across all regions in the new equilibrium. As shown in Table 4, Central Canada experiences an increase in its share of the Canadian population by around five percent at the expense of Western and Atlantic Canada. Although the smaller size of Canada relative to the U.S. implies that the effects are smaller for U.S. regions, the Mid West and South experience an increase and decrease in their population shares by around three and two percent respectively.

Region	Relative Own	Relative	Relative
	Trade Share	Welfare	Pop. Share
	π'_{nn}/π_{nn}	V_n'/V_n	λ'_n/λ_n
Atlantic Canada	0.6714	1.1828	0.9548
Central Canada	0.6051	1.1828	1.0526
Western Canada	0.7160	1.1828	0.8990
Mid West	0.9225	1.0179	1.0317
North East	0.9324	1.0179	1.0214
South	0.9756	1.0179	0.9790
West	0.9771	1.0179	0.9776

Table 4: Counterfactuals for Welfare under Population Mobility

Note: The table reports the results from a counterfactual in which international trade costs between Canadian and U.S. regions are reduced by 25 percent while intra-national trade costs between regions within each country remain unchanged; variables in the counterfactual equilibrium are denoted by a prime; variables in the initial equilibrium are denoted without a prime. See the appendix for definitions of U.S. and Canadian regions.

The effects of external trade liberalization between countries are therefore shaped to a quantitatively relevant extent by internal reallocations of resources across regions within countries. These intra-national reallocations of resources provide a distinct mechanism through which trade affects welfare that is often neglected in studies of international integration. Incorporating this mechanism changes the welfare gains from trade, which can no longer be inferred from the change in each region's trade share with itself but also depend on changes in the allocation of the mobile factor. While geographical mobility ensures that the welfare gains from trade for the mobile factor are equalized across regions, the resulting changes in the distribution of the mobile factor imply uneven changes in the price of the immobile factor.

7 Conclusions

Much trade occurs intra-nationally across regions within countries, where not only goods but also factors of production are mobile across locations. This paper extends a recent class of quantitative models of trade to incorporate factor mobility. This class includes models of constant returns to scale with no agglomeration forces (including Eaton and Kortum 2002 augmented with factor mobility) and models featuring increasing returns to scale and agglomeration forces (including a multi-region version of Helpman 1998). Although we model factor mobility between an arbitrary number of regions connected by an arbitrary pattern of geographical trade frictions, and allow for factor mobility across either all or a subset of regions, the analysis remains highly tractable.

We show that the standard conditions for general equilibrium in this class of models are augmented by an additional relationship that determines the distribution of the mobile factor within countries. Using the structure of the model and observed data on factor supplies and trade shares, we show that the dispersion of mobile factors can be decomposed into the contributions of locational fundamentals, market access and their covariance. We generalize the standard approach to undertaking counterfactuals in quantitative trade models to allow for such factor mobility. The key new feature of this generalized approach is that external trade liberalization leads to endogenous reallocations of resources across regions within countries. We show that these endogenous reallocations have ramifications for the representative agent's welfare gains from trade. A location's trade share with itself is no longer a sufficient statistic for the welfare gains from trade, which also depend on endogenous changes in the distribution of mobile factors. We derive a bias correction term for the standard measure of a country's aggregate welfare gains from trade that adjusts for these endogenous changes in the distribution of mobile factors within countries. In a quantitative application to trade between Canadian provinces and U.S. states, we show that these endogenous reallocations of resources within countries are quantitatively relevant for evaluating the representative agent's welfare gains from trade.

A Appendix

A.1 Proof of Proposition 1

To establish the proposition, we proceed in two steps. First, we show that given the population share for each location (λ_n) the wage equation (15) and trade share equation (16) determine a unique equilibrium wage in each location (w_n) and hence a unique equilibrium trade share (π_{ni}). Second, we use the population mobility condition (17) together with the wage equation (15) and trade share equation (16) to establish the existence of an equilibrium population share (λ_n).

In our first step, we use the trade share (16) in the wage equation (15) to obtain:

$$w_i \lambda_i = \sum_{n \in \mathbb{N}} \frac{T_i \left(d_{ni} w_i \right)^{-\theta}}{\sum_{k \in \mathbb{N}} T_k \left(d_{nk} w_k \right)^{-\theta}} w_n \lambda_n.$$
(68)

Given population shares in each location (λ_n), the above wage equation provides a system of N equations in the N unknown wages for each location n = 1, ..., N. Note that this wage equation (68) takes the same form as equation (3.14) on page 1734 of Alvarez and Lucas (2007). Using (68), we can define the following excess demand system:

$$\Xi(\mathbf{w}) = \frac{1}{w_i} \left[\sum_{n \in N} \frac{T_i \left(d_{ni} w_i \right)^{-\theta}}{\sum_{k \in N} T_k \left(d_{nk} w_k \right)^{-\theta}} w_n \lambda_n - w_i \lambda_i \right] = 0,$$
(69)

where **w** denotes the vector of wages across locations.

Lemma 1 Given population shares in each location (λ_n) , there exists a unique wage vector $\mathbf{w} \in \Re_{++}^n$ such that $\Xi(\mathbf{w}) = 0$.

Proof. Note that $\Xi(\mathbf{w})$ has the following properties:

(i) $\Xi(\mathbf{w})$ is continuous.

(ii) $\Xi(\mathbf{w})$ is homogeneous of degree zero.

(iii) $\mathbf{w} \cdot \Xi(\mathbf{w}) = 0$ for all $\mathbf{w} \in \Re_{++}^n$ (Walras Law)

(iv) There exists a s > 0 such that $\Xi_i(\mathbf{w}) > -s$ for each location *i* and all $\mathbf{w} \in \Re_{++}^n$

(v) If $\mathbf{w}^m \to \mathbf{w}^0$ where $\mathbf{w}^0 \neq 0$ and $w_i^0 = 0$ for some country *i*, then

$$\max_{j}\left\{\Xi_{j}\left(\mathbf{w}^{m}\right)\right\}\to\infty$$

(vi) Gross substitutes:

$$\begin{aligned} \frac{\partial \Xi_i\left(\mathbf{w}\right)}{\partial w_j} &> 0 \quad \text{for all } i, j, i \neq j, \text{ for all } \mathbf{w} \in \Re_{++}^n, \\ \frac{\partial \Xi_i\left(\mathbf{w}\right)}{\partial w_i} &< 0 \quad \text{for all } i, \text{ for all } \mathbf{w} \in \Re_{++}^n, \end{aligned}$$

from which the proposition follows. See Propositions 17.B.2, 17.C.1 and 17.F.3 of Mas-Colell, Whinston and Green (1995) and Theorems 1-3 of Alvarez and Lucas (2007). ■

Our second step uses the population mobility condition together with the comparative statics of the wage and trade equations. The population mobility condition (17) can be re-written as follows:

$$\lambda_n = \frac{(T_n)^{\frac{\alpha}{\theta(1-\alpha)}} H_n}{(T_n)^{\frac{\alpha}{\theta(1-\alpha)}} H_n + \sum_{k \neq n} \left(\frac{\pi_{nn}}{\pi_{kk}}\right)^{\frac{\alpha}{\theta(1-\alpha)}} (T_k)^{\frac{\alpha}{\theta(1-\alpha)}} H_k}.$$
(70)

Suppose that $\lambda_n \to 0$ while $\lambda_k > 0$ for $k \neq n$. As $\lambda_n \to 0$, the left-hand side of the population mobility condition (70) converges towards zero. Additionally, as $\lambda_n \to 0$ and $\lambda_k > 0$ for $k \neq n$, the wage equation (15) and trade equation (16) imply that $w_n = \infty$ and $\pi_{nn} = 0$, while w_k and π_{kk} take finite positive values for $k \neq n$. Using these results in the population mobility condition (70), the right-hand side of this condition converges towards infinity as $\lambda_n \to 0$ and $\lambda_k > 0$ for $k \neq n$.

In contrast, suppose that $\lambda_n \to 1$ and $\lambda_k \to 0$ for $k \neq n$. As $\lambda_n \to 1$, the left-hand side of the population mobility condition (70) takes a finite positive value. Additionally, as $\lambda_n \to 1$ and $\lambda_k \to 0$ for $k \neq n$, the wage equation (15) and trade equation (16) imply that w_n and π_{nn} take finite positive values, while $w_k = \infty$ and $\pi_{kk} = 0$ for $k \neq n$. Using these results in the population mobility condition (70), the right-hand side of this condition converges towards zero as $\lambda_n \to 1$ and $\lambda_k \to 0$ for $k \neq n$.

Combining these two sets of results, the left-hand side of the population mobility condition (70) is strictly less than the right-hand side as $\lambda_n \rightarrow 0$ and is strictly greater than the right-hand side as $\lambda_n \rightarrow 1$. Furthermore, this result holds for each location $n \in N$. Noting that the population mobility condition (70), wage equation (15) and trade equation (16) are continuous in $\{\lambda_n, w_n, \pi_{nn}\}$ for all $n \in N$, it follows that there exists an equilibrium allocation of population shares across locations $\{\lambda_n\}$ for which the left and right-hand sides of the population mobility condition (70) are equalized for each location $n \in N$. This allocation involves strictly positive population shares $\lambda_n > 0$ for all locations for which T_n , $H_n > 0$. Given this equilibrium allocation of population shares $\hat{\lambda}_n$, the wage equation (15) and trade equation (16) determine equilibrium wages for each location (\hat{w}_n) and hence equilibrium trade shares ($\hat{\pi}_{ni}$) as established in Lemma 1.

A.2 Special case of Two Regions

In this section of the appendix, we provide a formal analysis of the special case of two regions shown in Figure 1. In this case, the wage equation (15) becomes:

$$\lambda_{A} = \pi_{AA}\lambda_{A} + \pi_{BA}w_{B}(1 - \lambda_{A}),$$

$$w_{B}(1 - \lambda_{A}) = \pi_{AB}\lambda_{A} + \pi_{BA}w_{B}(1 - \lambda_{A}),$$
(71)

where have chosen the wage in region A as the numeraire ($w_A = 1$). The trade equation (16) is now:

$$\pi_{ni} = \frac{T_i (d_{ni} w_i)^{-\theta}}{\sum_{k \in \{A,B\}} T_k (d_{nk} w_k)^{-\theta}}, \qquad n, i \in \{A, B\},$$
(72)

while the population mobility condition (17) can be written as:

$$\lambda_A = \frac{(T_A)^{\frac{\alpha}{\theta(1-\alpha)}} H_A}{(T_A)^{\frac{\alpha}{\theta(1-\alpha)}} H_A + \left(\frac{\pi_{AA}}{\pi_{BB}}\right)^{\frac{\alpha}{\theta(1-\alpha)}} (T_B)^{\frac{\alpha}{\theta(1-\alpha)}} H_B}.$$
(73)

The existence of equilibrium can be established by exactly the same line of reasoning as for an arbitrary number of regions in the previous section. First, for given population shares in each location $\{\lambda_A, 1 - \lambda_A\}$, there exist unique wages $\{1, w_B\}$ that solve the wage equation (71) and trade equation (72). Second, equilibrium population shares are determined by combining the population mobility condition (73) with the wage equation (71) and trade equation (72).

Following the same argument as in the previous section, as $L_A \rightarrow 0$, the left-hand side of the population mobility condition (73) is strictly less than the right-hand side. In contrast, as $\lambda_A \rightarrow 1$, the left-hand side of the population mobility condition (73) is strictly greater than the right-hand side. Noting that the population mobility condition (73), wage equation (15) and trade equation (16) are continuous in $\{\lambda_n, w_n, \pi_{nn}\}$ for $n \in \{A, B\}$, it follows that there exists an equilibrium allocation of population shares across the two locations $\{\lambda_A, 1 - \lambda_A\}$ for which the left and right-hand sides of the population mobility condition (73) are equalized. Finally, with two regions, the wage equation (71) and trade equation (72) imply that w_B and π_{AA} are monotonically increasing in λ_A , while π_{BB} is monotonically decreasing in λ_A . It follows that the right-hand side of the population mobility condition (73) as shown graphically in Figure 1.

A.3 Proof of Proposition 4

The proof of the proposition again proceeds in two steps. First, we show that given the population share for each location (λ_n) the wage equation (32) and trade share equation (33) determine a unique equilibrium wage in each location (w_n) and hence a unique equilibrium trade share (π_{ni}). Second, we use the population mobility condition (34) together with the wage equation (32) and trade share equation (33) to establish the existence of an equilibrium population share (λ_n).

In our first step, we use the trade share (33) in the wage equation (32) to obtain

$$w_i \lambda_i = \sum_{n \in \mathbb{N}} \frac{\lambda_i \left(d_{ni} w_i \right)^{1-\sigma}}{\sum_{k \in \mathbb{N}} \lambda_k \left(d_{nk} w_k \right)^{1-\sigma}} w_n \lambda_n$$

Note that this wage equation takes the same form as equation (3.14) on page 1734 of Alvarez and Lucas (2007). Given population shares in each location (λ_n), there exists a unique wage vector $\mathbf{w} \in \Re_{++}^n$ such that $\Xi(\mathbf{w}) = 0$ by the same arguments as in the proof of Lemma 1 in Section A.1 above.

Our second step uses the population mobility condition together with the comparative statics of the wage and trade equations. The population mobility condition (34) can be re-written as follows:

$$\lambda_{n} = \frac{H_{n}^{\frac{(1-\alpha)(\sigma-1)}{\sigma(1-\alpha)-1}}}{H_{n}^{\frac{\sigma(1-\alpha)(\sigma-1)}{\sigma(1-\alpha)-1}} + \sum_{k \neq n} H_{k}^{\frac{(1-\alpha)(\sigma-1)}{\sigma(1-\alpha)-1}} \left(\frac{\pi_{nn}}{\pi_{kk}}\right)^{\frac{\alpha}{\sigma(1-\alpha)-1}}}.$$
(74)

Since we focus on parameter values for which $\sigma(1 - \alpha) > 1$, equilibrium population shares can be determined using exactly the same line of reasoning as in Section A.1. Taking the limits $\lambda_n \rightarrow 0$ and $\lambda_n \rightarrow 1$ and using continuity establishes the existence of an equilibrium allocation of population shares across locations for which the left and right-hand side of the population mobility condition (74) are equalized for each location $n \in N$. This equilibrium allocation involves strictly positive population shares $\lambda_n > 0$ for all locations for which $H_n > 0$. Given this equilibrium allocation of population shares $\hat{\lambda}_n$, the wage equation (32) and trade equation (33) determine equilibrium wages for each location (\hat{w}_n) and hence equilibrium trade shares ($\hat{\pi}_{ni}$) as discussed above.

A.4 Proof of Proposition 7

The proof of the proposition again proceeds in two steps. First, we show that given the population share for each location n in each country j (λ_n^j) the wage equation (51), price index equation (53) and trade share equation (52) determine a unique equilibrium wage (w_n) and price index (P_n) in each location and hence a unique equilibrium trade share (π_{ni}). Second, we use the population mobility condition (54) together with the wage equation (51), price index equation (53) and trade share equation (52) to establish the existence of an equilibrium population share (λ_n^j).

In our first step, we use the trade share (52) in the wage equation (51) to obtain:

$$\begin{split} w_{i}\lambda_{i}^{A}\bar{L}^{A} &= \sum_{n\in N^{A}} \left[\frac{T_{i}\left(d_{ni}w_{i}^{\beta+\eta}P_{i}^{1-\beta-\eta}\left(\lambda_{i}^{A}\right)^{\eta}\left(\frac{L^{A}}{H_{i}}\right)^{\eta}\right)^{-\theta}}{\sum_{k\in N^{A}}T_{k}\left(d_{nk}w_{k}^{\beta+\eta}P_{k}^{1-\beta-\eta}\left(\lambda_{k}^{A}\right)^{\eta}\left(\frac{L^{A}}{H_{k}}\right)^{\eta}\right)^{-\theta} + \sum_{k\in N^{B}}T_{k}\left(d_{nk}w_{k}^{\beta+\eta}P_{k}^{1-\beta-\eta}\left(\lambda_{k}^{B}\right)^{\eta}\left(\frac{L^{B}}{H_{k}}\right)^{\eta}\right)^{-\theta}} \right] w_{n}\lambda_{n}^{A}\bar{L}^{A} \\ &+ \sum_{n\in N^{B}} \left[\frac{T_{i}\left(d_{ni}w_{i}^{\beta+\eta}P_{k}^{1-\beta-\eta}\left(\lambda_{k}^{A}\right)^{\eta}\left(\frac{L^{A}}{H_{k}}\right)^{\eta}\right)^{-\theta} + \sum_{k\in N^{B}}T_{k}\left(d_{nk}w_{k}^{\beta+\eta}P_{k}^{1-\beta-\eta}\left(\lambda_{k}^{B}\right)^{\eta}\left(\frac{L^{B}}{H_{k}}\right)^{\eta}\right)^{-\theta}} \right] w_{n}\lambda_{n}^{B}\bar{L}^{B}, \\ w_{i}\lambda_{i}^{B}\bar{L}^{B} &= \sum_{n\in N^{A}} \left[\frac{T_{i}\left(d_{ni}w_{k}^{\beta+\eta}P_{k}^{1-\beta-\eta}\left(\lambda_{k}^{A}\right)^{\eta}\left(\frac{L^{A}}{H_{k}}\right)^{\eta}\right)^{-\theta} + \sum_{k\in N^{B}}T_{k}\left(d_{nk}w_{k}^{\beta+\eta}P_{k}^{1-\beta-\eta}\left(\lambda_{k}^{B}\right)^{\eta}\left(\frac{L^{B}}{H_{k}}\right)^{\eta}\right)^{-\theta}} \right] w_{n}\lambda_{n}^{A}\bar{L}^{A} \\ &+ \sum_{n\in N^{B}} \left[\frac{T_{i}\left(d_{ni}w_{k}^{\beta+\eta}P_{k}^{1-\beta-\eta}\left(\lambda_{k}^{A}\right)^{\eta}\left(\frac{L^{A}}{H_{k}}\right)^{\eta}\right)^{-\theta} + \sum_{k\in N^{B}}T_{k}\left(d_{nk}w_{k}^{\beta+\eta}P_{k}^{1-\beta-\eta}\left(\lambda_{k}^{B}\right)^{\eta}\left(\frac{L^{B}}{H_{k}}\right)^{\eta}\right)^{-\theta}} \right] w_{n}\lambda_{n}^{B}\bar{L}^{B}, \end{split}$$

where $\{\bar{L}_A, \bar{L}_B\}$ are the exogenous labor endowments of each country.

Note that this wage equation for each country takes the same form as equation (3.14) on page 1734 of Alvarez and Lucas (2007). Given population shares in each location (λ_n^j for $j \in \{A, B\}$), there exists a unique wage vector $\mathbf{w} \in \Re_{++}^n$ such that $\Xi(\mathbf{w}) = 0$ by the same arguments as in the proof of Lemma 1 in Section A.1 above.

Our second step uses the population mobility condition together with the comparative statics of the wage and trade equations. The population mobility condition (54) can be re-written as follows:

$$\lambda_{n}^{A} = \frac{(T_{n})^{\overline{\theta(\beta+\eta-\alpha\beta)}}H_{n}}{(T_{n})^{\overline{\theta(\beta+\eta-\alpha\beta)}}H_{n} + \sum_{k \neq n, k \in N^{A}} \left(\frac{\pi_{nn}}{\pi_{k}}\right)^{\overline{\theta(\beta+\eta-\alpha\beta)}}(T_{k})^{\overline{\theta(\beta+\eta-\alpha\beta)}}H_{k}}, \qquad n \in N^{A},$$

$$\lambda_{n}^{B} = \frac{(T_{n})^{\overline{\theta(\beta+\eta-\alpha\beta)}}H_{n}}{(T_{n})^{\overline{\theta(\beta+\eta-\alpha\beta)}}H_{n} + \sum_{k \neq n, k \in N^{B}} \left(\frac{\pi_{nn}}{\pi_{kk}}\right)^{\overline{\theta(\beta+\eta-\alpha\beta)}}(T_{k})^{\overline{\theta(\beta+\eta-\alpha\beta)}}H_{k}}, \qquad n \in N^{B},$$
(75)

Equilibrium population shares across locations *n* within each country $j \in \{A, B\}$ can be determined using exactly the same line of reasoning as in Section A.1. Taking the limits $\lambda_n^j \to 0$ and

 $\lambda_n^j \to 1$ for each country $j \in \{A, B\}$ and using continuity establishes the existence of an equilibrium allocation of population shares across locations for which the left and right-hand side of the population mobility condition (75) are equalized for each location $n \in N^j$ within each country j. This equilibrium allocation involves strictly positive population shares $\lambda_n^j > 0$ for all locations for which $T_n, H_n > 0$. Given this equilibrium allocation of population shares $\hat{\lambda}_n^j$, the wage equation (51) and trade equation (52) determine equilibrium wages for each location (\hat{w}_n) and hence equilibrium trade shares ($\hat{\pi}_{ni}$) as discussed above.

A.5 Data on Canadian and U.S. Regions

The source for the trade data between Canadian and U.S. regions for 1993 is Anderson and van Wincoop (2003). The sources for the population data for Canadian and U.S. regions for 1990 are Statistics Canada and the U.S. Census Bureau. To ensure positive bilateral trade flows between all locations, we aggregate Canadian provinces and U.S. states to the level of statistical regions, which are defined below. We also exclude Northern Canada, which accounts for a small share of Canadian trade and population.

Table 5: Canadian provinces and regions.

	<u> </u>	/	
Atlantic Canada	Central Canada	Western Canada	
New Brunswick	Ontario	Alberta	
Newfoundland and Labrador	Quebec	British Columbia	
Nova Scotia		Manitoba	
Prince Edward Island		Saskatchewan	

		0		
North East	Mid West	South	West	
Connecticut	Indiana	Delaware	Arizona	
Maine	Illinois	District of Columbia	Colorado	
Massachusetts	Michigan	Florida	Idaho	
New Hampshire	Ohio	Georgia	New Mexico	
New Jersey	Wisconsin	Maryland	Montana	
New York	Iowa	North Carolina	Utah	
Pennsylvania	Kansas	South Carolina	Nevada	
Rhode Island	Minnesota	Virginia	Wyoming	
Vermont	Missouri	West Virginia	Alaska	
	Nebraska	Alabama	California	
	North Dakota	Kentucky	Hawaii	
	South Dakota	Mississippi	Oregon	
		Tennessee	Washington	
		Arkansas	C	
		Louisiana		
		Oklahoma		
		Texas		

References

- Allen, Treb and Costas Arkolakis (2013) "Trade and the Topography of the Spatial Economy," NBER Working Paper, 19181.
- [2] Alvarez, Fernando and Robert E. Lucas (2007) "General Equilibrium Analysis of the Eaton-Kortum Model of International Trade," *Journal of Monetary Economics*, 54, 1726-1768.
- [3] Anderson, James and Eric van Wincoop (2003) "Gravity with Gravitas: a Solution to the Border Puzzle," *American Economic Review*, 93(1), 170-192.
- [4] Arkolakis, Costas, Arnaud Costinot and Andrés Rodriguez-Clare (2012) "New Trade Models, Same Old Gains," American Economic Review, 102(1), 94-130.
- [5] Armington, P. S. (1969) "A Theory of Demand for Products Distinguished by Place of Production," IMF Staff Papers, 16, 159-178.
- [6] Blanchard, Olivier and Larry Katz (1992) "Regional Evolutions," Brookings Papers on Economic Activity, 23(1), 1-76.
- [7] Bound, John and Harry J. Holzer (2000) "Demand Shifts, Population Adjustments and Labor Market Outcomes During the 1980s," *Journal of Labor Economics*, 18(1), 20-54.
- [8] Caliendo, Lorenzo and Fernando Parro (2012) "Estimates of the Trade and Welfare Effects of NAFTA," NBER Working Paper, 18508.
- [9] Coşar, Kerem and Pablo Fajgelbaum (2013) "Internal Geography, International Trade, and Regional Specialization," *NBER Working Paper*, 19697.
- [10] Costinot, Arnaud, Donaldson, Dave and Ivana Komunjer (2012) "What Goods Do Countries Trade? A Quantitative Exploration of Ricardo's Ideas," *Review of Economic Studies*, 79(2), 581-608.
- [11] Courant, Paul N. and Alan V. Deardorff (1992) "International Trade with Lumpy Countries," *Journal of Political Economy*, 100(1), 198-210.
- [12] Courant Paul N. and Alan V. Deardorff (1993) "Amenities, Nontraded Goods, and the Trade of Lumpy Countries," *Journal of Urban Economics*, 34(2), 299-317.
- [13] Davis, Morris A. and François Ortalo-Magne (2011) "Household Expenditures, Wages, Rents," *Review of Economic Dynamics*, 14(2) 248-261.
- [14] Davis, Donald and David Weinstein (2002) "Bones, Bombs, and Break Points: The Geography of Economic Activity," American Economic Review, 92(5), 1269-1289.

- [15] Dekle, Robert, Eaton, Jonathan and Kortum, Samuel (2007) "Unbalanced Trade," American Economic Review, 97(2), 351-355.
- [16] Desmet, Klaus and Esteban Rossi-Hansberg (2014) "Spatial Development," American Economic Review, forthcoming.
- [17] Donaldson, Dave (2014) "Railroads of the Raj: Estimating the Impact of Transportation Infrastructure," American Economic Review, forthcoming.
- [18] Eaton, Jonathan and Kortum, Samuel (2002) "Technology, Geography, and Trade," *Econometrica*, 70(5), 1741-1779.
- [19] Eaton Jonathan, Sam S. Kortum and Francis Kramarz (2011a) "An Anatomy of International Trade: Evidence from French Firms," *Econometrica*, 1453-1498.
- [20] Eaton, Jonathan, Sam Kortum, Brent Neiman and John Romalis (2011b) "Trade and the Global Recession," NBER Working Paper, 16666.
- [21] Eaton, Jonathan, Samuel Kortum and Sebastian Sotelo (2012) "International Trade: Linking the Micro and the Macro," NBER Working Paper, 17864.
- [22] Fieler, Cecilia (2011) "Non-Homotheticity and Bilateral Trade: Evidence and a Quantitative Explanation," *Econometrica*, 79(4), 1069-1101.
- [23] Fujita, Masahisa, Paul Krugman and Anthony J. Venables (1999) The Spatial Economy: Cities, Regions and International Trade, Cambridge: MIT Press.
- [24] Hanson, Gordon H. (1996) "Localization Economies, Vertical Organization, and Trade," American Economic Review, 86(5), 1266-78.
- [25] Hanson, Gordon H. (1997) "Increasing Returns, Trade, and the Regional Structure of Wages," *Economic Journal*, 107, 113-133.
- [26] Hanson, Gordon H. (2005) "Market Potential, Increasing Returns, and Geographic Concentration," *Journal of International Economics*, 67(1), 1-24.
- [27] Head, Keith and John Ries (2001) "Increasing Returns versus National Product Differentiation as an Explanation for the Pattern of U.S.-Canada Trade," *American Economic Review*, 91, 858-876.
- [28] Helpman, Elhanan (1998) "The Size of Regions," In Topics in Public Economics: Theoretical and Applied Analysis, ed. David Pines, Efraim Sadka, and Itzhak Zilcha, Cambridge: Cambridge University Press,
- [29] Hillberry, Russell and David Hummels (2008) "Trade Responses to Geographic Frictions: A Decomposition Using MicroData," *European Economic Review*, 52, 527-550.

- [30] Hsieh, Chang-Tai and Ralph Ossa (2011) "A Global View of Productivity Growth in China," NBER Working Paper, 16778.
- [31] Krugman, Paul (1991a) "Increasing Returns and Economic Geography," Journal of Political Economy, 99(3), 48399.
- [32] Krugman, Paul (1991b) Geography and Trade, Cambridge MA: MIT Press.
- [33] Markusen, James R. (1983) "Factor Movements and Commodity Trade as Complements," Journal of International Economics, 14(34), 341-356.
- [34] Mas-Collel, Andreu, Michael D. Whinston and Jerry R. Green (1995) *Microeconomic Theory*, Oxford: Oxford University Press.
- [35] McCallum, John (1995) "National Borders Matter: Canada-U.S. Regional Trade Patterns," American Economic Review, 85(3), 615-623.
- [36] Mundell, R. A. (1957) "International Trade and Factor Mobility," American Economic Review, 47, 321-337.
- [37] Jones, Ron (1967) "International Capital Movements and the Theory of Tariffs and Trade," Quarterly Journal of Economics, 81, 1-38.
- [38] Ossa, Ralph (2011) "Trade Wars and Trade Talks with Data," NBER Working Paper, 17347.
- [39] Ramondo, Natalia, Andrés Rodríguez-Clare and Milagro Saborio (2012) "Scale Effects and Productivity Across Countries: Does Country Size Matter?," NBER Working Paper, 18532.
- [40] Rauch, James E. (1991) "Comparative Advantage, Geographic Advantage, and the Volume of Trade," *Economic Journal*, 101, 1230-1244.
- [41] Redding, Stephen J. and Daniel M. Sturm (2008) "The Costs of Remoteness: Evidence from German Division and Reunification," *American Economic Review*, 98(5), 1766-1797.
- [42] Redding, Stephen J. and Anthony J. Venables (2004) "Economic Geography and International Inequality," *Journal of International Economics*, 62(1), 53-82.
- [43] Rossi-Hansberg, Esteban (2005) "A Spatial Theory of Trade," American Economic Review, 95(5), 1464-1491.
- [44] Simonovska, Ina and Michael E. Waugh (2013) "The Elasticity of Trade: Estimates and Evidence," *Journal of International Economics*, 92(1), 34-50.
- [45] Valentinyi, Akos and Berthold Herrendorf (2008) "Measuring Factor Income Shares at the Sectoral Level," *Review of Economic Dynamics*, 11, 820-835.